

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Malik, Muhammad Adeel; Ali Mirza, Rinchan; Rehman, Faiz Ur

Working Paper Frontier rule and conflict

WIDER Working Paper, No. 2023/128

**Provided in Cooperation with:** United Nations University (UNU), World Institute for Development Economics Research (WIDER)

*Suggested Citation:* Malik, Muhammad Adeel; Ali Mirza, Rinchan; Rehman, Faiz Ur (2023) : Frontier rule and conflict, WIDER Working Paper, No. 2023/128, ISBN 978-92-9267-436-6, The United Nations University World Institute for Development Economics Research (UNU-WIDER), Helsinki, https://doi.org/10.35188/UNU-WIDER/2023/436-6

This Version is available at: https://hdl.handle.net/10419/283824

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# WIDER Working Paper 2023/128

# Frontier rule and conflict

Adeel Malik,<sup>1</sup> Rinchan Ali Mirza,<sup>2</sup> and Faiz Ur Rehman<sup>3</sup>

October 2023

United Nations University World Institute for Development Economics Research

wider.unu.edu

Abstract: We examine whether frontier rule, which disallows frontier residents from a recourse to formal institutions of conflict management and disproportionately empowers tribal elites, provides a more fragile basis for maintaining social order in the face of shocks. Combining a historical border separating frontier from non-frontier regions in north-western Pakistan with 10km-by-10km grid cell-level data on conflict in a spatial regression discontinuity design framework, we show that areas that historically fell under frontier rule experienced significantly higher violence against the state after 9/11. We argue that the 9/11 tragedy represented a shock to grievances against the state which, in the absence of formal avenues of conflict management, led to a sharp surge in attacks against state targets in frontier areas. We show that the surge in 'sovereignty-contesting' forms of violence in these regions was partly carried out through the systematic assassination of tribal elites who were the main pillar of frontier rule that guaranteed social order. In our empirical analysis we rule out several important competing explanations behind the post-9/11 rise in violence in frontier areas, including the possibility of conflict spilling over from Afghanistan, income shocks (proxied by military operations), and drone attacks.

Key words: institutions, historical frontiers, conflict

#### JEL classification: D02, D74, N45, P48

**Acknowledgements:** This research has been supported by UNU-WIDER as part of its 'Institutional legacies of conflict' project. For their constructive feedback we are also grateful to participants at the LSE Economic History Seminar, the Development Economics Workshop at University of Kent, the UNU-WIDER Development Conference (2022) in Helsinki, the XIXth World Economic History Congress (2022) in Paris, the Pathways to Development Conference (2022) at LUMS, the AALIMS (2023) conference at Stanford University, and the Applied Development Economics (2023) conference at Lahore School of Economics. The usual disclaimers apply.

Note: tables at the end of the paper, before the appendix

This study is a significantly revised version of 'Frontier governmentality', available as WIDER Working Paper 2023/42.

This study has been prepared within the UNU-WIDER project Institutional legacies of violent conflict.

Copyright © UNU-WIDER 2023

UNU-WIDER employs a fair use policy for reasonable reproduction of UNU-WIDER copyrighted content—such as the reproduction of a table or a figure, and/or text not exceeding 400 words—with due acknowledgement of the original source, without requiring explicit permission from the copyright holder.

Information and requests: publications@wider.unu.edu

ISSN 1798-7237 ISBN 978-92-9267-436-6

https://doi.org/10.35188/UNU-WIDER/2023/436-6

Typescript prepared by Lesley Ellen and Siméon Rapin.

United Nations University World Institute for Development Economics Research provides economic analysis and policy advice with the aim of promoting sustainable and equitable development. The Institute began operations in 1985 in Helsinki, Finland, as the first research and training centre of the United Nations University. Today it is a unique blend of think tank, research institute, and UN agency—providing a range of services from policy advice to governments as well as freely available original research.

The Institute is funded through income from an endowment fund with additional contributions to its work programme from Finland and Sweden, as well as earmarked contributions for specific projects from a variety of donors.

Katajanokanlaituri 6 B, 00160 Helsinki, Finland

The views expressed in this paper are those of the author(s), and do not necessarily reflect the views of the Institute or the United Nations University, nor the programme/project donors.

<sup>&</sup>lt;sup>1</sup>University of Oxford and Oxford Centre for Islamic Studies, Oxford, UK, adeel.malik@qeh.ox.ac.uk; <sup>2</sup>University of Kent, Canterbury, UK, r.a.mirza@kent.ac.uk; <sup>3</sup>Institute of Business Administration, Karachi, Pakistan, faizrehman@iba.edu.pk

"To speak of frontier governmentality in the modern world, then, is to speak of a long history of violence." (Benjamin Hopkins, Ruling the Savage Periphery: Frontier Governance and the Making of the Modern State, 2020: 194)

#### 1 Introduction

The great imperial powers ruled their frontier territories differently from the rest of their colonial domains, creating a 'rule of difference' that was manifested in distinct administrative, legal, and institutional practices. Described as 'frontier governmentality', these practices 'constituted a discrete form of rule unique to frontier spaces' (Hopkins 2020: 17).<sup>1</sup> In these regions, described by historians as 'liminal spaces' or 'edges of empires', the state had a relatively thin presence, and social order was almost entirely predicated on elite intermediation. Frontier residents were thus treated as 'imperial objects' who remained susceptible to arbitrary state action, which contrasted sharply with non-frontier regions where colonial subjects had access to a full panoply of state institutions. The upshot was that frontier communities were 'encapsulated' in their own local traditions and 'enclosed', in institutional terms, from the rest of the colony (Hopkins 2020). Originally introduced in the late nineteenth and earlier twentieth centuries, frontier governance quickly became 'an administrative archetype, widely replicated the world over' (Hopkins 2020: 3–4). In his magnum opus, *Ruling the Savage Periphery: Frontier Governance and the Making of the Modern State*, Hopkins (2020: 6) alludes to the 'near simultaneous construction of a system of frontier administration on a cosmopolitan canvas'.<sup>2</sup>

Frontier governmentality persisted long after the end of colonial rule and served as a powerful, yet understudied, institutional legacy.<sup>3</sup> It resulted in long-term economic and political marginalization of frontier territories and shaped the conditions for social order (Hopkins 2015, 2020; Naseemullah 2022). Recent work by historians and political scientists shows that the 'highly uneven territorialisation of power' (Naseemullah 2022: xi) through which the frontier was created as a 'spatial, political, and administrative category' (p. 384) left a profound historical legacy for explaining conflict against the state (Kolsky 2015; Hopkins 2020; Tripodi 2020).<sup>4</sup> Three interrelated features of frontier rule make it prone to conflict. First, given that state authority was imperfectly penetrated in frontier regions, local populations had, at best, only a tenuous and indirect link with the state, thereby shaping state–society relationships in profound ways. Second, frontier residents did not have recourse to institutions of conflict management (e.g., courts, electoral politics) that were typically available to other colonial subjects. Instead, traditional modes of dispute resolution supervised by local elites (chiefs and tribal elders) were officially sanctioned and adopted by the imperial administration. Third, frontier rule represented a highly personalized

<sup>&</sup>lt;sup>1</sup> Such institutional heterogeneity is rooted in different colonial motivations. In regions where i) colonizers faced serious external threats, ii) the ability to extract resources was limited, and iii) the relative costs of extending state authority were high, they tended to delegate greater authority to local elites and established more exceptional institutional arrangements (Naseemullah 2022).

<sup>&</sup>lt;sup>2</sup> There are several prominent historical illustrations of this exceptional frontier rule in colonial territories, such as the North-West Frontier of British India (Frontier Crimes Regulation 1887), Kenya's northern frontier with Somaliland (Special Districts Administrative Ordinance of 1934), India's North-East (Chin Hill Regulations of 1896), and Iraq's Basra Vilayet (Tribal Civil and Criminal Disputes Regulation of 1915).

<sup>&</sup>lt;sup>3</sup> Modern states have continued to follow the colonial practice of governing their frontier regions through exceptional institutional arrangements (e.g., negotiated security guarantees with local elites, reliance on non-state security actors, legal adjudication through tribal elites rather than courts, dependence on local norms, and informal institutions).

<sup>&</sup>lt;sup>4</sup> As Naseemullah (2022: 17) argues, the 'spatial framework of governance diversity, with roots in colonial rule and post-colonial politics, represents the key to understanding the politics of conflict'.

form of rule that empowered the 'man on the spot' (Hopkins 2020: 23). While colonial authorities routinely used local leaders and chiefs as 'mediators' and 'conduits', frontier governmentality constituted a specific 'sub-category' of indirect rule that delegated even greater power to local elites, sharing with them the state's power over coercion and social control (Mamdani 1999; Naseemullah 2022; Naseemullah and Staniland 2016).<sup>5</sup>

In this paper we argue that, while this elite-negotiated frontier rule can ensure social order for extended periods of time, it is more susceptible to violence in the face of disruptive shocks. The lower resilience of frontier rule in the face of shocks stems from there being fewer avenues for conflict management, lower trust in the state, and disproportionately greater reliance on elite intermediation.<sup>6</sup> An intrinsic feature of frontier governmentality is its 'fluid and unstable character', which is 'subject to constant processes of bargaining between different parties'. Even the very idea of 'statehood' is 'negotiable' and typically 'mediated' through local elites (Cuvelier et al. 2014: 346). When elites face a threat of elimination in the face of shocks, the entire social order begins to unravel, leading to what Naseemullah (2022) describes as 'sovereignty-contesting' forms of violence manifested in local insurgencies in frontier regions. Such violence is more likely to:

[...] arise and persist when there are fewer institutionalized resources that might draw different groups into competition over them, and the relative absence of interpenetration between society and the state that would institutionalize this competition. Instead, groups violently reject the legitimacy of the state and its ability to organize relations. (Naseemullah 2022: 20–21)

To examine the impact of frontier governmentality on conflict, we study a setting in British India's North-West Frontier that has long been considered as an archetypical case of exceptional frontier rule. Here, the British imperial administration devised radically different institutional arrangements for governing the frontier tribes as opposed to the settled regions. While colonial subjects in settled regions had access to a full panoply of institutions, including courts, police, civil bureaucracy, and electoral institutions, frontier tribes were enclosed in a state of institutional 'exception' where the routine institutions of conflict management were absent. Pursuing what was essentially a 'laissez faire policy of administration', the colonial state delegated its sovereignty to local elites who were crucial to both dispute resolution and securing the frontier (Tripodi 2009). Conflict resolution depended on the intervention of the council of tribal elders, popularly known as the jirga, and operated under the direct oversight of colonial bureaucracy. Alongside, tribal elites were used to recruit a local militia (known as *khassadars*) for securing the border. In return for their support, elites received a subsidy from the colonial state. This 'markedly different' colonial rule had a 'defining effect' on state-society relationship, laying the basis for legal and political marginalization in frontier regions (Hopkins 2015: 380). Frontier residents existed on the 'margins of the state, excluded from the national body politics and defined by an era of colonial governance with limited rights and access to judicial systems' (Siddiqui 2018).7 This exceptional rule had a 'lasting postcolonial afterlife' as it remained in operation from 1901 to 2018 when Pakistan's tribal agencies in Federally Administered Tribal Areas (FATA) were officially merged with neighbouring settled districts (Hopkins 2015: 385).

<sup>&</sup>lt;sup>5</sup> This entailed, among others, the power to recruit local militias and maintain security in their local jurisdictions.

<sup>&</sup>lt;sup>6</sup>We are partly motivated here by Rodrik (1999) who demonstrated the importance of formal 'institutions of conflict management' in stabilizing economic growth in the face of external shocks.

<sup>&</sup>lt;sup>7</sup> From the perspective of frontier populations, frontier rule resulted in both exclusion and dependence. It not only excluded local populations from having recourse to formal institutions under colonial rule (e.g., bureaucracy and courts) but also subjected them to a tight 'economic squeeze', which made them dependent on the colonial economy.

The legacies of this frontier rule (henceforth, FR) in what is now Pakistan's north-west, has been extensively studied by historians and anthropologists (Allen 2012; Beattie 2013; Kolsky 2015; Hopkins 2015, 2020; Pant 2018). In this paper, we provide the first systematic attempt at probing its impact on conflict against the state. To do so, we exploit a highly fine-grained database on conflict incidents for the period, 1970–2018, at the 10km-by-10km grid cell level. We then utilize historical sources to construct the frontier border demarcating the sharp institutional discontinuity between FR and non-FR areas that has remained in force since 1901. Combining these two elements in a spatial regression discontinuity (SRD) design, we compare observations on both sides of the border to test whether historical exposure to frontier rule predicts contemporary conflict. Our results show that, on average, between 1970 and 2018 areas that fell just inside the FR border. Concretely, individuals in grid cells just inside the FR border were 57 per cent more exposed to conflict incidents against the state than similar individuals just outside the FR border.

We recognize potential threats to our identification strategy, including the possibility of discontinuities in key environmental and structural characteristics in closely situated spatial units that are omitted from our analysis. These could be potentially correlated with the emergence of FR and have influenced the post-FR trajectory of conflict. Reassuringly, we do not find any statistically significant discontinuity across a wide array of geographic and climatic factors, including ruggedness, slope, topography, precipitation, temperature, and wheat suitability. We also demonstrate that historical factors, such as pre-colonial conflict and population density, also vary smoothly across the FR border. To ensure that our results are not simply capturing a pure *border effect*, we successively exclude grid cells located close to the FR border and show that our core empirical discontinuity remains.

Our findings emanate from a relatively stringent identification strategy that restricts the sample to grid cells within a 50-km buffer zone of the FR border and controls for 20-km border segment fixed effects. This effectively means that we compare observations within a 50-km buffer zone along the FR border within the same 20-km border segment. Our results hold up to a battery of robustness checks, which we report in the appendix. These include: changing the size of the buffer zone (40 km and 60 km instead of 50 km); altering the size of the border segments (18 km and 15 km instead of 20 km); using an alternative database on conflict; using alternative functional forms for the running variable; choosing other manually chosen bandwidths; applying a different kernel weighting strategy for observations close to the FR border; and shifting the original FR border south-westward (i.e. further inland) as a placebo exercise.

Probing the temporal dimension, we show that the effect of frontier rule on conflict was only activated after the invasion of Afghanistan by the United States (US) in 2001. We show that prior to 2001 there were no systematic differences in conflict against the state between FR and non-FR regions. However, FR regions witnessed a sharper spike in conflict shortly after 2001 relative to geographically proximate non-FR regions. We argue that the year 2001 represented a universal shock to the Pakistani population's grievance against the state for its decision to lend support to the US-led 'war on terror' in the aftermath of 9/11. However, this rise in anti-state sentiment was particularly pronounced in FR areas where, in addition to the state's support for the war on terror, the Pakistani military increased its presence under US pressure for the purposes of patrolling the Pak-Afghan border and 'dominating the space' (Nawaz 2011: 8). In our causal chain, this led to a discontinuous rise in sovereignty-contesting forms of violence in FR areas where formal institutions of conflict management were largely absent, there was an overreliance on elites for dispute resolution, and trust in the state was historically low. Figure 1 provides a brief sketch of our argument and the mechanisms at work.

#### Figure 1: The transmission mechanism from frontier rule to conflict



Source: authors' construction.

To understand why 9/11 evoked a discontinuous response in FR areas, we provide several pieces of evidence that are consistent with our mechanism. First, we estimate linear probability models using data from a representative household-level survey and demonstrate that individuals residing in FR areas were less likely to consider members of the national legislature as the main recourse for dispute resolution. Instead, frontier residents were significantly more reliant on the assembly of tribal leaders (*jirga*) for adjudicating disputes. Second, we show that frontier residents also had a significantly lower trust in formal state institutions (e.g. parliament and courts). Finally, we document the unravelling of social order through the systematic targeting of tribal leaders in FR regions, which was a key component of the local insurgents' strategy in perpetrating violence against the state. Moving across the border from non-FR to FR areas, we find a discontinuous rise in attacks against tribal elders. In a context where the local social order primarily hinged on elite intermediation, elimination of these elites represented an important strategy adopted by non-state actors, which created an institutional vacuum and exposed the FR areas to greater violence against the state.

Our findings on the greater vulnerability of FR-governed areas to conflict in the wake of external shocks are consistent with prior qualitative research (Naseemullah 2014, 2022). Specifically, our results shed empirical light on the claim in Naseemullah (2014: 519) that 'the destruction of the regular means through which social elites were able to negotiate with the agents of the state was a significant enabling factor for insurgent conflict in the tribal northwest'. We also engage with other competing explanations and rule out the potential role of conflict spillover from neighbouring Afghanistan, the post-9/11 income shock induced by military operations, US drone strikes, and the differential provision of public infrastructure (roads, railroads, waterways, and health centres).

Our paper contributes to a well-traversed academic terrain on the long-run determinants of conflict, especially the role of institutions (Bang and Mitra 2017; Bellows and Miguel 2006; Chauvin 2016; Ciccone 2008; Collier and Hoeffler 1998, 2004, 2007; Fearon 2007; Fearon and Laitin 2003; Heldring 2021; Herbst 2000; La Ferrara and Bates 2001; Miguel et al. 2004; Sambanis 2005; Skaperdas 2008; Voors and Bulte 2014) (see Blattman and Miguel (2010) for a detailed review).<sup>8</sup> Our work differs from existing literature in several important respects.

<sup>&</sup>lt;sup>8</sup> Conflict is typically described as the result of bargaining failure and commitment problems. To enforce commitments over time, it is important to have strong government institutions and checks and balances on executive power. Weak

First, while prior scholarship has begun to characterize the effect of broad institutional characteristics, we have limited knowledge of the role of 'specific political and legal institutions' in driving conflict (Blattman and Miguel 2010).<sup>9</sup> A related challenge emanating from the overwhelmingly cross-country focus of prior work is that institutional heterogeneity within countries remains understudied. To this end our contribution is to focus on a specific form of indirect rule which created distinct institutional arrangements for governing frontier regions and whose effects remain largely unexplored. By doing so our study addresses an important research question highlighted by Sambanis (2005): Does the extent to which a state exercises control over its peripheral territories impact conflict? While there is a vibrant literature analysing frontier governance arrangements in political science, international relations, anthropology, and area studies (Colona and Jaffe 2016; Meagher 2012; Meagher et al. 2014; Roitman 2005), their role in driving conflict is largely ignored in the mainstream economics literature. As Cuvelier et al. (2014: 346–47) note, the 'evidence base' for past studies is limited, and, while the literature offers 'new analytical tools', there are few notable attempts to 'systematically gather empirical evidence'.

Second, as Blattman and Miguel (2010: 30) argue, 'an important limitation' of prior work is that it mainly focuses on domestic drivers of conflict. The role of international factors has only recently come under closer intellectual scrutiny with studies probing the impact of global institutional transformations (Wimmer and Min 2006), terms of trade shocks (Frankema et al. 2018), conflict spillovers from neighbouring countries (Hegre and Sambanis 2006), presence of transboundary ethnic groups (Gleditsch 2007), refugee flows (Salehyan and Gleditsch 2006), foreign aid (Nielsen et al. 2011; Savun and Tirone 2011), and foreign interventions (Aydin 2012). However, barring these exceptions, the salience of international issues has been insufficiently explored. We contribute to this literature by adding an important nuance which revolves around the role of geopolitical shocks (i.e. the post-9/11 war on terror) in shaping the relationship between institutions and conflict. In this regard, our work provides confirmatory evidence on two important insights in Blattman and Miguel (2010: 18), namely that: (a) the impact of institutions on conflict can be conditional on other factors and (b) 'non-economic explanations' such as citizen's 'emotional and ideological outrage' can shape grievances that underpin violent action.

Third, a key empirical challenge for conflict studies is to isolate the impact of institutions from other powerful drivers of conflict, such as geography, climate, and income shocks. While prior work is mostly cross-country in nature, Blattman and Miguel (2010: 26) recommend that 'future empirical work should achieve more credible causal inference by focusing on a single, or a small number of, exogenous conflict determinants'. In this spirit our paper is situated within a growing genre of studies that exploit sub-national (or sub-regional) variations in conflict while focusing on a core conflict driver and using a robust empirical strategy. For example, Dell and Querubin (2018) exploit discontinuities in US military strategies across neighbouring regions in Vietnam to study their differential impact on insurgency and local governance. Moscona et al. (2020) exploit ethnic boundaries in sub-Saharan Africa to empirically demonstrate a connection between segmentary lineage organization and conflict across ethnic groups. Similarly, Michalopolous and Pappaioannou (2016) investigate the impact of the colonial-era border-making which partitioned ethnicities in Africa on contemporary conflict.<sup>10</sup> An advantage of our empirical setting is that many of the

state capacity has also emerged as an important correlate of violence (Herbst 2000; Bates 2008; Besley and Persson 2008, 2010).

<sup>&</sup>lt;sup>9</sup> In fact, as Blattman and Miguel (2010: 28) argue, 'several of the institutional characteristics have yet to be carefully defined and measured', and there is a need to 'disaggregate' and unpack the institutional effect.

<sup>&</sup>lt;sup>10</sup> Using a carefully crafted identification strategy, they find that ethnicities whose traditional homelands were partitioned as a consequence of drawing up colonial borders during the 'Scramble for Africa' experienced significantly

dimensions traditionally considered as important for explaining conflict, such as ethnicity, religion, and segmentary lineage, are naturally controlled for in the case of our frontier border.

Finally, our work is situated within the expanding literature on the long-run impact of history on development—see Nunn (2009, 2021) for exhaustive reviews—especially the persistent impact of colonial-era institutions (Acemoglu et al. 2001; Acemoglu and Robinson 2012; Engerman and Sokoloff 1994). Our paper contributes to this literature by highlighting that colonialism was not a 'singular treatment', as colonizers tended to build diverse institutional arrangements even within the same territory (Mamdani 2018; Boone 2003). Furthermore, besides showing that historical institutions matter for development, we also demonstrate 'why' and 'when' they matter. Our work thus feeds into a select literature in historical political economy that emphasizes the role of 'time-varying persistence', which suggests that the impact of history can remain latent for a long time until it is activated through interaction with other factors or shocks (Belmonte and Rochlitz 2019; Cantoni et al. 2019; Fouka and Voth 2013; Ochsner and Roesel 2017).

Within the broader genre of historical work, we contribute more directly to two key sub-strands. The first is the work by Acemoglu et al. (2014), which probes the impact of the power and authority of African chiefs on development. While tribal elites are a central part of our story, our story is not one of extractive elites but their role within the larger architecture of frontier governance. We show how the oversized role of local elites can render the underlying institutional order more vulnerable to violence in the face of shocks. Second, we complement prior scholarship on the impact of historical borders (Becker et al. 2016; Dell 2010). In particular, our paper connects more directly with an emerging strand of scholarship probing the impact of historical frontiers on longrun inequality (Oto-Peralías and Romero-Ávila 2017), economic geography (Chronopoulos et al. 2021), and individual and gender norms (Bazzi, Fiszbein, and Gebresilasse 2020; Bazzi, Brodeur, Fiszbein, and Haddad 2023). Our work complements this scholarship by treating frontiers not just as a purely geographic dimension or distinctive norm-creating spaces but as a profound institutional category. To this end, our paper is closer in spirit to Popescu (2023) that examines the developmental legacy of a military buffer zone in Hapsburg Empire. We differ from this work in two key respects. Firstly, the nature of our institutional treatment is different. While Popescu (2023) focuses on the state's limited provision of infrastructure and institutional arrangements for property and labour markets, we highlight the role of institutions of conflict management.<sup>11</sup> Secondly, we focus on a different outcome of interest (i.e. conflict against the state). We thus bring fresh insights to a highly niche literature on historical frontiers by providing what is, in our humble assessment, the first empirical attempt at examining the impact of historically embedded institutional differences in frontier regions on contemporary conflict.<sup>12</sup>

higher levels of civil conflict in the long run. They attribute the persistently higher levels of civil conflict associated with partitioned ethnic groups to the greater likelihood of these groups engaging in armed conflict to counter the state's repression.

<sup>&</sup>lt;sup>11</sup> In fact, we demonstrate that there are no statistically significant differences in the provision of public infrastructure in closely situated spatial units in FR and non-FR areas (see Section 5.4). Another point of difference is the role of elites. In Popescu's frontier setting the task of securing the frontier was delegated to peasants and refugees while in our context, there was a greater delegation of power to local elites.

<sup>&</sup>lt;sup>12</sup> Our paper complements Callen et al. (2020) who study Pakistan's north-western frontier as a case study to explain why states leave their territories ungoverned. Our research also feeds into an expanding literature in political science and development studies on hybrid governance in fragile states (Arnaut et al. 2008; Boege et al. 2008; Cleaver 2002; Garrett et al. 2009; Hagmann and Péclard 2010; Lund 2006; Menkhaus 2006; Raeymaekers et al. 2008). While this literature offers useful typologies, thick descriptions, and contextual analysis, our work provides a more rigorous empirical focus.

The remainder of this paper is structured as follows. Section 2 sets out the historical background of this study. Section 3 describes the data and methods. Section 4 presents the main findings of our empirical analysis. Section 5 identifies plausible mechanisms. Section 6 concludes.

# 2 Context

In this section we discuss the emergence of frontier rule (FR) in the north-western frontier of British India, highlighting both its salient features and the imperial rationale behind its introduction. We also discuss the salience of the 9/11 shock for explaining violence against the state in frontier regions.

# 2.1 Frontier rule in British India

The north-western frontier of British India was carved out of Punjab as a separate province in 1901 by Lord Curzon, the viceroy of British India (1898–1905). The newly created North-West Frontier Province was bifurcated into settled regions and the frontier tracts. The latter were populated by tribes adjacent to Afghanistan.<sup>13</sup> While the two regions shared the same ethnicity, religion, and social structure, they were subjected to a sharply distinctive category of institutional rule by the British.<sup>14</sup> The cartographic divide between frontier and non-frontier areas signified a sharp institutional discontinuity. As Sir Olaf Caroe, a colonial-era governor of the frontier province, aptly noted: 'the line of administration stopped like a tide almost at the first contour' of frontier territories (Caroe 1958: 239). In frontier tracts, this was manifested in relatively thin state presence and more exceptional institutional arrangements that excluded frontier inhabitants from the paraphernalia of colonial institutions that were otherwise available to colonial subjects in settled districts. For example, while the inhabitants of British India in non-frontier regions had at least some modicum of judicial protection through divisional and district courts along with the universal application of the Indian Penal Code, frontier dwellers were 'legally disenfranchised' (Hopkins 2020).

An important manifestation of this was the creation of an alternative judicial system for frontier tribes which relied on local elite intermediation (and adjudication) rather than formal colonial institutions, such as the courts and judiciary.<sup>15</sup> Local tribal elites (e.g., *Maliks* and *Khans*) were empowered to adjudicate disputes through customary practices and informal institutions known as the *jirga* (a traditional congregation of tribal elders where decisions were made by consensus). While *jirgas* were informal consultative bodies that historically varied in form and purpose, the British standardized the modes of tribal governance and made *jirgas* the principal avenue for

<sup>&</sup>lt;sup>13</sup> Initially, the NWFP province consisted of the following five settled districts: Peshawar, Kohat, Hazara, Bannu, and Dera Ismail Khan. Frontier areas consisted of the tribal agencies that lied in the liminal spaces between the settled districts and Durand line boundary with Afghanistan agreed in 1893. These were: Dir, Swat, Chitral, Bajaur, Mohmand, Khyber, Kurram, Orakzai, North Waziristan, and South Waziristan tribal agencies.

<sup>&</sup>lt;sup>14</sup> It is important to note that Pashtun tribes existed on both sides of the frontier rule border. The difference between frontier and non-frontier areas was thus not merely a distinction between tribal and non-tribal. The emphasis on border tribes was more a part of colonial rhetoric and imagination. As Scott (2009: 30) argues, there is a tendency for centralized state to view 'all those who had a reason to flee state power for whatever reason' as 'tribalizing themselves'. As Scott further argues: 'Ethnicity and tribe began, by definition, where sovereignty and taxes ended'.

<sup>&</sup>lt;sup>15</sup> An important legal instrument associated with this exceptional legal regime was the Frontier Crimes Regulation (FCR), which was first introduced in 1872. The FCR codified pre-existing practices of collective punishments and has consistently been described as 'draconian' and discriminatory, as it gave the accused in frontier areas neither a right to appeal nor equality before law (Mahsud et al. 2021).

dispute resolution. The tribal judiciary, described as the 'Council of Elders' in the 1887 regulation, was typically composed of a handful of appointed tribal leaders. Importantly, these leaders did not necessarily act as extractive elites and operated within an inherently egalitarian and non-hierarchical Pashtun society.<sup>16</sup>

Frontier tribes also had distinct security and policing arrangements. Compared to the elaborate structure of policing in place for settled districts, the British administration relied on locally recruited militias (*khassadars* and levies) for maintaining law and order in tribal jurisdictions. The *khassadars* possessed the powers to arrest and, if needed, refer the offenders to tribal councils for dispensing justice. From 1915 they were also assisted by a paramilitary force known as the Frontier Constabulary (FC) that was also responsible for guarding and patrolling the border between the tribal agencies and settled districts and watching over foreign threats. However, the FC only had the power to monitor, sanction, and keep a general watch. It neither had traditional policing powers nor full coercive monopoly (Hopkins and Marsden 2012; Naseemullah 2014). Instead, army officials could be attached to local paramilitary forces and support from regular Army troops stationed in neighbouring cantonments could also be sought. The British Indian Army also maintained garrisons in strategic locations in tribal areas, such as Khyber and Waziristan.

Frontier areas were also weakly penetrated by state bureaucracy. While the settled districts were administered through a well-developed bureaucratic system that included deputy commissioners and revenue officials, colonial interests in frontier areas were primarily overseen by political agents. These were officers of the British Raj who helped to maintain relations with tribes on behalf of the colonial state, gathered intelligence, and provided financial subsidies to tribal elites. <sup>17</sup> Lastly, a key feature of frontier rule was the complete absence of political representation. Under the British rule, restricted enfranchisement was gradually extended to residents of settled districts through legislative councils, assemblies, and district and municipal boards. However, even this nominal and limited electoral representation remained unavailable to frontier residents. These historically embedded institutional differences are summarized in Table 1.

The fundamental architecture of frontier rule survived long after the British departed from India. After gaining independence in 1947, frontier residents benefited from better provision of infrastructure and job opportunities in the public sector. <sup>18</sup> However, the colonially sanctioned 'rule of difference' continued unabated. Frontier regions remained legally and politically marginalized. The 1956 Constitution of Pakistan maintained the application of Frontier Crimes Regulations in FR areas. This effectively meant that neither the Supreme Court nor High Courts were able to exercise their jurisdiction in frontier regions. Furthermore, the parliamentary laws that governed the operation of political parties did not extend to frontier areas. Consequently, neither of Pakistan's mainstream political parties were able to operate and field candidates in these regions. Limited adult franchise was extended in 1997 whereby tribal leaders were given the right to vote in a non-party-based election. While the exclusionary legal and political restrictions applied to these

<sup>&</sup>lt;sup>16</sup> Colonial archives consistently describe the border tribes as 'exceedingly democratic' with 'no recognized headmen' (Ibbetson 1881: 10). In this context, the basis of authority rests on consensual legitimacy. As Ibbetson (1881: 201) notes: 'Each section of a tribe, however small, has its leading Khan Khel or Chief House, usually the eldest branch of the tribe, whose Malik is known as Khan, and acts as chief of the whole tribe. But he is seldom more than their leader in war and their agent in dealings with others; he possesses influence rather than power; and the real authority rests with the *jirgah*, a democratic council composed of all the Maliks'.

<sup>&</sup>lt;sup>17</sup> Naseemullah (2014: 511) describes the role of political agents as 'part military commander, part diplomat, and part spy'. Financial inducements to tribal elites typically consisted of annual allowances paid by the colonial administration to tribal chiefs (Maliks) and stipends to recruit armed militias (Naseemullah 2014; Hopkins 2015).

<sup>&</sup>lt;sup>18</sup> As per the most recent census conducted in 2017, the frontier tribal agencies consist of around 3,000 villages with a total population of roughly three million people.

frontier regions were finally withdrawn in 2018, this has thus far only existed on paper as little has substantially changed on the ground (Mahsud et al. 2021).<sup>19</sup>

Historians have long argued that the markedly different institutional arrangements that have governed frontier tribes represent both a profound and persistent colonial inheritance of British rule in India (Embree 1977; Nichols 2001; Tanguay-Renaud 2002; Stewart 2007; Hopkins 2020). This set the two regions on very different paths of state–society relationships, with the state being a more distant and alien construct in frontier areas. According to Naseemullah (2022), it is precisely these pre-configured and historically embedded differences in state capacities and state–society relationships that are important predictors of contemporary conflict, especially insurgency-based violence at the local level.

#### Imperial rationale for frontier rule

In extending 'exceptional' legal, administrative, and political arrangements in frontier territories, colonial rulers were influenced by three factors: fear, greed, and frugality (Naseemullah 2022). In their frontier domains, colonizers typically faced an external threat, had limited prospects for resource extraction, and faced relatively high costs of extending state authority. The north-west frontier of British India has historically acted as a strategic frontier nestled between Afghanistan, Central Asia, China, and present-day Pakistan (formerly British India). The frontier areas effectively represented a buffer zone for the British Empire against possible external threats from Tsarist Russia in the context of the geo-political competition between great powers, often referred to as the 'Great Game' (Becker 2012; Davis 1926; Hopkirk 2001).<sup>20</sup> However, the possibility of a Russian over-stretch became less likely over time, thereby reducing the seriousness of the Russian threat. Instead, as Tripodi (2009: 3) has argued, the 'true threat to British India was perceived to lie within'. Regular skirmishes between colonial forces and frontier tribes posed continuing challenges to state authority in frontier areas, which could have potentially generated a more 'popular uprising' and undermined British rule in the rest of India.

Another important consideration was the relatively limited tax potential in frontier agencies. Compared to settled districts, frontier areas had on average different climate and topography, potentially resulting in more limited agricultural surplus.<sup>21</sup> Consequently, for a more frugal colonial dispensation, it was relatively costly to extend state authority by establishing formal institutional structures (Callen et al. 2020). Frontier institutional arrangements might also have been shaped by other factors, such as the differential experience of conflict and development in pre-colonial times. Our spatial regression discontinuity (SRD) design will therefore explicitly account for such historic, geographic, and environmental factors by showing that these features vary smoothly across the border.

<sup>&</sup>lt;sup>19</sup> While various official commissions were set up to review FR after independence, they only resulted in minor amendments. Similarly, senior members of the judiciary repeatedly passed critical judgements against FR in several high-profile cases. Prominent efforts in this regard include: the Quetta and Kalat Laws Commission, the Law Reform Commissions of 1958 and 1967–70, and the FATA Reforms Commission of 2005. None of these resulted in a substantive change.

<sup>&</sup>lt;sup>20</sup> The 'Great Game' signifies the geo-political competition during the nineteenth and early twentieth centuries between the Russian and British Empires for the control of Afghanistan, Central Asia, and neighbouring South Asian territories.

<sup>&</sup>lt;sup>21</sup> The settled or non-frontier districts consisted of fertile valleys, low hills, and plains. By contrast, the frontier tracts were mostly hilly terrains interspersed with small basins and valleys.

#### 2.2 The 9/11 shock to grievance against the state

The year 2001 represents a major turning point in international relations that had grave implications for localized conflict in Pakistan's frontier areas. The tragedy of 9/11 prompted United States (US) to initiate a major military response in the guise of the war on terror, triggering an attack against Afghanistan. As a pivotal neighbouring state, the US sought Pakistan's intelligence and logistical support to facilitate the Afghan war operations and persuaded the Pakistani military to move its troops close to the Afghan border. This entailed a major policy shift. Two decades earlier Pakistan's support was similarly critical in the Afghan Mujahideen's armed resistance against the Soviet Union, which was tacitly backed by the US and allied Western powers (Coll 2004; Dorril 2002; Khan 2013; Pentz 1987; Riedel 2014; Weinbaum 1991). At that time, the main objective was to free Afghanistan from Soviet occupation. Militant Islamic groups were mobilized and used as proxies against the Soviets, then considered as an anti-Islamic force. There was a broader convergence at the time between Pakistani state and frontier tribesmen in their opposition to Russian invasion. However, when in 2001 the US demanded renewed cooperation from Pakistan, it essentially entailed a major reversal of policy that the Pakistani establishment had pursued for the previous 20 years. Specifically, Pakistan was made to relinquish ties with the Taliban, many of whom had previously fought against the Soviets, and to support US war efforts against the group in Afghanistan. This was a 180-degree shift in Pakistan's Afghan policy, which had a deep institutional and ideological support base in the country.

In this context, the US attack on Afghanistan and the almost-overnight foreign policy shift by Pakistan's military ruler Parvez Musharraf were deeply unpopular.<sup>22</sup> The resentment against General Musharraf's decision to join the US-led war was both universal, affecting all regions in Pakistan, and persistent. This is reflected in a series of surveys and opinion polls. Opinion data from a Pew survey indicates that, while a majority of the Pakistani population (58 per cent) maintained an 'unfavourable' view of the US in 2000, this proportion increased to 82 per cent after 9/11.<sup>23</sup> Similarly, a Gallup poll conducted on 18 June 2002 showed that 62 per cent of Pakistanis were opposed to the government's decision to lend support to the US-led war on Afghanistan. Eleven years later results from a similar poll indicated that 71 per cent of respondents disapproved of cooperation with the US in its war on terror. The 9/11-induced grievance against the state was noticeably higher among residents of FR areas where, in response to the US demand, the military substantially increased its presence for securing the Afghan border.<sup>24</sup> While initially the military only 'dominated the space' without conducting 'aggressive operations', its mere presence was seen as an 'alien force doing the bidding of a foreign power' which fuelled a 'local uprising' (Nawaz 2011:6–8).<sup>25</sup>

The year 2001 thus represented a universal shock to grievance against the state, which disproportionately affected the frontier areas. This external shock was relatively exogenous to local conflict intensity along Pakistan's North-West Frontier. While recognizing that these regions had

<sup>&</sup>lt;sup>22</sup> The US attack on Afghanistan was perceived as an aggression by the world's dominant superpower against one of the poorest states. General Musharraf's regime tried to justify this policy shift on grounds of necessity and a survival option in the face of the shifting tide of global opinion and intense US pressure. For further details, see Reuters (2006).

<sup>&</sup>lt;sup>23</sup> *Global Attitudes Survey*, Pew Research Center. The data can be accessed at: https://www.pewresearch.org/global /datasets/.

<sup>&</sup>lt;sup>24</sup> A survey report showed that some 85 per cent of respondents in frontier areas opposed the presence of the US military in the region (Shinwari 2012: 86). In 2011 around 58 per cent of survey respondents viewed the US in 'very unfavourable terms' (Shinwari 2012: 129).

<sup>&</sup>lt;sup>25</sup> Nawaz (2011: 10) describes this policy of 'dominating space' as 'sitzkrieg', which meant 'sitting in camps without any aggressive actions'.

hosted many fighters involved in the armed struggle against the Soviets in the 1980s and that militants in Afghanistan might have maintained some links across the border in Pakistan after the Soviet withdrawal in 1989, the 9/11 attacks were part of a broader global phenomenon of Islamic Jihad that cannot be conceivably linked to the potential for conflict in frontier areas. Specifically, both the timing and location of the 9/11 shock are orthogonal to the prospects for localized conflict in frontier regions.

Pakistan witnessed a substantial surge in violence in the post-9/11 period. Around 12,891 attacks were conducted against the state and civilian targets after 2001, a noticeable increase compared to the 1,845 attacks in the entire pre-9/11 era stretching from 1970 to 2000 (GTD 2021). The intensity of such violence was considerably higher in frontier regions. While militant outfits involved in this violence primarily emerged after 9/11 (e.g. TTP and TNSM<sup>26</sup>), they benefited from the organizational infrastructure put together during the 1980s Afghan Jihad against the Soviets. Their recruits enjoyed deep social and ethnic ties with local populations and were overwhelmingly drawn from disenchanted segments of local tribes. These included, for example, the Alizai clan of Mehsuds, Ibrahim Khel clan of Utmanzai Waziris, Kaka Khel subtribe of Ahmadzai Waziris, and Mezi sub-tribe of the Zadranis (Jones and Fair 2010).

The post-9/11 acceleration of violence triggered a domestic and global policy response. Specifically, the FR areas witnessed military offensives, both by Pakistani and US forces, against alleged sanctuaries of terrorist outfits. From 2004 onwards, the US also carried out drone strikes against alleged Al-Qaeda-linked affiliates in Pakistan's North-West (Ahmed et al. 2019; Mahmood and Jetter 2019). These attacks increased after 2008, reached their peak in 2010, and declined precipitously thereafter until they were phased out by the Obama administration in 2016 (see Figure B1 in Appendix B). Drone attacks further intensified anti-US sentiment and have remained deeply unpopular across Pakistan.

Growing violence in frontier areas also prompted the military to launch selected offensives against local insurgents. However, trained in conventional warfare, the Pakistani military found itself unprepared for counter-insurgency operations after 2001. This had two key implications. First, the Pakistani military had to rely on the US for technical and financial assistance to carry out these offensives. This was seen as yet more evidence of collaboration with a foreign power and therefore further ignited local grievances. Second, earlier military operations were met with limited success. Described as achieving 'low' levels of success, these operations were inconclusive in that they only 'temporarily' cleared the violence-hit areas (Jones and Fair 2010: 68). The 'short-term successes' meant that military operations did not decisively hurt the insurgents' 'ability to attack the government' (Jones and Fair 2010: 70). However, successive operations did increase the military's capability, as reflected in the more effective *Zarb-e-Azb* military offensive in the North Waziristan Agency in 2014. The operation overlapped with the attack on Army Public School in Peshawar city in December 2014, which generated a nation-wide political resolve to fight terrorism. *Zarb-e-Azb* has been described as 'biggest and most well-coordinated operation' against local insurgents to date (Rehman et al. 2017).

<sup>&</sup>lt;sup>26</sup> TTP stands for 'Tehreek-e-Taliban Pakistan' and TNSM denotes 'Tehreek-e-Nifaze-Shariate Mohammdi'.

# 3 Data and variables

This section describes the data used to construct the main variables used in our empirical analysis. The unit of observation for our analysis is the 10km-by-10km grid cell.

# 3.1 Conflict data

The main source of our conflict data is the Global Terrorism Database (GTD 2021) that provides information on more than 200,000 conflict incidents across the world since 1970. For each incident, information is provided on the time (day, month, and year), location (latitude and longitude), fatalities (wounded and killed), type (assassination, explosion, suicide, hijacking, etc.), target (military, civilians, businesses, government officials, religious institutions, non-governmental organizations, etc.), the source (militant outfit that carried out the attack), and the motivation for the attack (political, religious, etc.). The GTD reports more than 14,500 conflict incidents in Pakistan from 1 January 1970 to 31 December 2018. Given our interest in explaining sovereignty-contesting forms of violence, our main outcome of interest captures attacks against military personnel and installations.<sup>27</sup> We describe these as attacks against the state. Using the GTD, we construct three measures of attacks against the state: (a) the number of incidents, (b) the number of deaths, and (c) the number of injuries in a 10km-by-10km grid cell for the period, 1970–2018. In the extended analysis, we also use a broader measure of state targets that also includes attacks against police, government officials, and state utility installations.<sup>28</sup>

# 3.2 Frontier rule border

We use primary sources, principally colonial-era archives, to construct the historic border that separates frontier agencies from settled districts. The border effectively separated the five settled districts of Peshawar, Kohat, Hazara, Bannu, and Dera Ismail Khan from the seven frontier agencies of Dir, Swat, Chitral, Khyber, Kurram, North Waziristan, and South Waziristan. Originally coming into force in 1901 on the direction of the then governor general of India, Lord Curzon, this was more than just a 'cartographic distinction' and represented a sharply distinct category of institutional rule that consistently remained in force till 2018 (Kolsky 2015: 1225). To construct this historic border, we imported the shapefile for an administrative map of British India in 1901 into the Quantum Geographical Information Systems (QGIS) application and used its map making tools to extract the frontier border.<sup>29</sup>

# 3.3 Controls

The validity of our SRD estimation method requires that other relevant dimensions apart from the treatment vary smoothly across the FR border. While we are not able to rule out discontinuities in every conceivable dimension, we do present evidence that a whole range of factors that could plausibly be linked to conflict vary smoothly across the FR border. In this regard we compile data on the following geographic, climatic, and historic factors.

<sup>&</sup>lt;sup>27</sup> Our focus on military targets is motivated by the well-accepted fact that the Pakistani military is the locus of state power and is therefore the primary target for 'sovereignty-contesting' forms of violence. Other coercive actors of the state, such as the police, levies, and *kbassadars*, are locally recruited and are therefore viewed less as an alien force and more as part of local society.

<sup>&</sup>lt;sup>28</sup> Further details on the 'broader' measure of attacks against the state are provided in Appendix A.

<sup>&</sup>lt;sup>29</sup> The shapefile for the 1901 administrative map of British India was obtained from Professor James Fenske's website: https://warwick.ac.uk/fac/soc/economics/staff/jefenske/data/.

## 3.3.1 Geographic characteristics

Data on elevation is provided by the Shuttle Radar Topography Mission (version 2018). The elevation information is recorded in metres at the resolution of 30 arc seconds, which is approximately equivalent to a 1km<sup>2</sup> level near the equator. Using these data, we calculate the terrain ruggedness index, average slope index, and topographic position index at the 10km-by-10km grid cell level. <sup>30</sup> We use the terrain ruggedness index (TRI) which was originally devised by Riley et al. (1999) and further developed by Nunn and Puga (2012). This TRI is calculated as the square root of the sum of the squared differences in elevation between a central point and the eight adjacent points on a grid of 30 arc seconds. For this study we construct the average terrain ruggedness for each 10km-by-10km grid cell, with higher values indicating higher terrain ruggedness.

Our measure of slope is a measure of change in elevation across space. The slope index is the weighted average of differences between adjacent 30 arc-second elevation points, with the weights being the inverse of distances between the points. The topographic position index (TPI) is another measure of an area's elevation relative to its surroundings. It is calculated by subtracting the mean elevation of eight surrounding 30 arc-second elevation points from a central elevation point. Using this procedure, we construct the average TPI for each grid cell in our dataset, with higher values indicating more extreme topography. The data on wheat suitability comes from the Food and Agriculture Organization's Global Agro-Ecological Zones (FAO-GAEZ) dataset, which is available from GAEZ Data Portal (2012). We compute grid cell-level measures by averaging over raster points within each 10km-by-10km grid cell. Note that the wheat suitability data is based on the 'low-input' and 'rain-fed' parameters that closely proxy the historical conditions under which wheat was grown in Pakistan.

# 3.3.2 Climate

Data on precipitation is provided by the Global Climate Database created by Hijmans et al. (2005), which is available from WorldClim (2020). Along with monthly average rainfall the database also provides the long-run average for the years 1970–2000 in millimetres. We match the average rainfall between 1970 and 2000 to each 10km-by-10km grid cell to construct a measure of long-term average difference in precipitation levels on either side of the FR border. For temperature data we again use the Global Climate Database by Hijmans et al. (2005), which also contains information on average temperature in °C, both on a monthly basis and as the long-term average for the period 1970—2000. We merge each grid cell with the average temperature between 1970 and 2000 to capture the long-run effects of temperature on both sides of the FR border.

## 3.3.3 Historical factors

We account for several key historical dimensions that could have potentially shaped the institutional discontinuity around the FR border and the trajectory of conflict against the state. The first is the pre-colonial population density. Data on pre-FR (i.e. before 1901) population density is extracted from the HYDE (2006) database, which provides internally consistent 30 arc second  $(1 \text{km}^2)$  grid cell-level estimates of population density at 100-year intervals for the last 12,000 years. The second dimension is pre-colonial conflict. For this we utilize a dataset compiled by Dincecco et al. (2022) that draws on information on historical conflicts, organized alphabetically by individual conflict names, in a celebrated volume from Jaques (2007). For each individual conflict, Jaques provides a paragraph-length description on the type of conflict (e.g., land, sea,

<sup>&</sup>lt;sup>30</sup> Grid cell-level aggregates are extracted using the QGIS *Zonal Statistics* command which calculates various statistics like average, median, standard deviation, etc., for raster datasets in defined zones.

etc.), its date, approximate duration (e.g. single day), approximate location, and major participants of the conflict. For other dimensions like Mughal-era roads and Islamic trade and pilgrimage routes, we use geo-referenced data from the Old World Trade Routes Project (Ciolek 1999).

# 3.4 Attacks on tribal elders (Maliks)

To probe the mechanism through which FR affects conflict against the state in the post-9/11 period, we construct different measures of violence against tribal chiefs. We use geocoded information in the GTD database to construct a variable aggregating the number of attacks against tribal chiefs in each grid cell.<sup>31</sup> We also construct measures aggregating the number of deaths and number of injuries in attacks against tribal elders, both calculated at the grid cell-level.

# 3.5 Descriptive evidence

Table 2 reports average differences in each of the three measures of conflict against the state on either side of the FR border. The means for FR and non-FR areas are followed by the estimated difference between the two values for each of the measures. Columns 1–3 report averages for the whole sample, whereas columns 4–6 restrict the sample to a 50-km buffer zone on either side of the FR border. The latter of these is the main sample we use for our empirical analysis. Across all measures, and regardless of whether we use the full or restricted sample, the intensity of conflict against the state is statistically significantly higher in FR areas relative to non-FR areas. In particular, the FR areas have between 36 per cent and 57 per cent more conflict intensity than the non-FR areas based on the specific measure that is used. Figure 2 plots each conflict event between 1970 and 2018 on a district-level map of Pakistan's North-West Frontier using the latitude and longitude coordinates. As the figure shows, the density of attacks against the state is noticeably higher inside the FR boundary (left of the border). We will show later in this paper that this difference in the density of attacks primarily emerges during the post-9/11 period.

<sup>&</sup>lt;sup>31</sup>The first ever attack against a tribal elder in the GTD occurred in 2006. This means that the data on violent incidents against tribal elders comes entirely from the post-9/11 period.



Figure 2: Spread of conflict incidents against the state across the FR border (1970-2018)

Note: the map shows the spread of conflict incidents against the state from 1970 to 2018. Each black dot represents an attack against the state (defined here as military personnel or installations) and the red line denotes the historical border (as of 1901) separating frontier agencies (left side of the border) from settled districts (right side of the border).

Source: authors' compilation. This map is created from GTD (2021) data through the QGIS application.

#### 4 Spatial regression discontinuity estimates

#### 4.1 Identification strategy

Utilizing the sharp institutional discontinuity between areas that historically fell under frontier rule (FR areas) and settled regions (non-FR areas), we estimate a spatial regression discontinuity (SRD) specification that takes the following form:

$$y_{i,i} = \alpha_0 + \phi_i + \alpha_1 \cdot Frontier_Rule_i + X'_i\beta + f(geographic_location_i) + \epsilon_{i,i}$$
(1)

where  $y_{i,j}$  is a measure of conflict against the state in a 10km-by-10km grid cell *i* along the border segment *j*. Frontier\_Rule<sub>i</sub> is a dummy variable indicating that grid cell *i* is inside the FR boundary.  $X'_i$  is a vector of covariates that includes the following set of grid-cell-level geographic, climatic, and historic controls: terrain ruggedness, slope, topography, wheat suitability, temperature, precipitation, pre-FR conflict incidence, and pre-FR population density.  $f(geographic_location_i)$ is a polynomial that controls for a smooth function of the geographic location of grid cells. In our main analysis, we use a grid cell's Euclidean distance from the FR border as the running variable and, following Gelman and Imbens (2019), use a local linear specification which is estimated separately on both sides of the border. We also experiment with several other functional forms in our robustness analysis. We divide the FR boundary into fixed 20-km segments to which grid cells are then matched;  $\phi_j$  are the associated border segment fixed effects which ensure that grid cells are compared across the same segment of the border.<sup>32</sup> Finally, to account for spatial correlation, we cluster the standard errors at the FR border segment level. Our parameter of interest is  $\alpha_1$ , the SRD estimate of the impact of frontier rule on conflict against the state. As long as the determinants of unobservable traits (e.g., geography, climate, and history) vary smoothly across the FR border, this estimate can be interpreted as causal.

The unit of observation in our empirical analysis is a 10km-by-10km grid cell, and the sample is restricted to grid cells falling within a 50-km buffer zone around the FR border. Figure 3 illustrates our SRD set-up, showing contiguous 10km-by-10km grid cells on either side of a specific segment of the FR border (red line), with some areas falling under frontier rule (grey cells) and others outside it (white cells). It also displays the precise location of conflict incidents against the state (black dots) along the specific segment. Figure 4 shows the 50-km buffer zone (shaded in grey) within which our analysis is restricted.

# 4.2 Validity of the SRD design

The SRD approach allows us to compare conflict incidence in areas that are geographically sufficiently close to each other, but where one area is subject to frontier rule and the other is not.<sup>33</sup> A key assumption of the SRD approach is that unobservable factors vary smoothly across the FR border. If frontier areas had some innate unobservable characteristics that both increased their propensity for violence and determined the emergence of frontier rule during the colonial era, this could potentially bias our results. Such unobservable characteristics can originate from different sources grounded in the history, geography, and climate of the frontier areas. While it is impossible to directly test for this assumption, we can nevertheless provide strong evidence in favour of its validity by showing the absence of discontinuity at the FR border for a variety of geographic, climatic, and historic correlates of conflict. In doing so we closely follow prior literature relating conflict with such dimensions as ruggedness (Fearon and Laitin 2003; Nunn and Puga 2012), climate and topography (Miguel et al. 2004; Burke et al. 2015; Ivigun et al. 2017; Chambru 2019), historical population density (Herbst 2000; Reid 2012), and historical exposure to conflict (Fearon and Laitin 2014; Dincecco et al. 2022). These dimensions are either directly correlated with conflict or serve as proxies for conflict determinants that are difficult to observe. For example, low per capita income, negative income shocks, and poverty are consistently flagged as important correlates of conflict (Blattman and Miguel 2010). While historical (especially pre-colonial) data on income are unavailable, climatic factors (e.g. rainfall) and historical population density act as useful proxies. Similarly, state authority and capacity have been identified as important determinants of conflict (Besley and Reynal-Querol 2014; Depetris-Chauvin 2015; Wig 2016). Geographic factors that constrain the reach of the state (Scott 2009) can thus act as proxies for state authority and capacity, dimensions that are otherwise hard to measure.

To rule out such conflict determinants, we estimate Equation 1 with each of the relevant dimensions as the dependent variable and demonstrate that the resulting SRD estimates are statistically insignificant and do not highlight a discontinuity across the FR border. We show this for the following dimensions: terrain ruggedness, slope, topography, wheat suitability, temperature, precipitation (mean and standard deviation), pre-colonial conflict intensity, and population density.

<sup>&</sup>lt;sup>32</sup> Other key studies that incorporate border segment fixed effects in an SRD specification include Dell (2010), Dell et al. (2018), and Asher et al. (2022).

<sup>&</sup>lt;sup>33</sup> Our approach is therefore akin to past studies that have leveraged a similar spatial focus on a small sub-region (see, for example, Lee and Schultz 2012; Dell and Olken 2020; Lowes and Montero 2021; Popescu 2023).

Table 3 reports the estimates from a variant of Equation 1 which only includes the border segment fixed effects. For each of the eight factors, the coefficient estimate on the FR indicator variable is consistently small in magnitude and never statistically different from 0. Additionally, in Appendix B (Figure B2), we present visual evidence showing binscatter plots of the unconditional relationship between each of the six main factors and the distance from the FR border. Even in these raw data, no discontinuity is apparent in any of these factors. In fact, there is a substantial overlap in the 95 per cent confidence intervals on either side of the distance to the border cut-off.

Besides discontinuities in geographic, climatic, and historical factors, another potential concern is that other characteristics, such as social structure, ethnic composition, and religious identity can vary discontinuously at the FR border. For instance, prior work on conflict in Africa has shown that ethnic groups organized around segmentary lineages are more likely to engage in violent conflict that is both larger in scale and more prolonged in duration (Moscona et al. 2020). However, an advantage of focusing our empirical analysis on a narrow strip of North-West Pakistan is that the social structure and associated features are similar across both sides of the FR border and therefore naturally controlled for.



Figure 3: Illustration of the SRD set-up

Note: this figure illustrates the SRD set-up. Every square is a 10km-by-10km gridcell. The red line is the FR border. The grey shade represents the area inside the FR border, whereas the white shade represents area outside the FR border. The black dots represent geocoded locations of conflict incidents against the state.

Source: authors' construction. The map is created from GTD (2021) data through QGIS software.

#### Figure 4: Area of study for the SRD set-up



Note: this figure shows the area of study within which our sample is restricted. It encompasses all those grid cells that lie within a 50-km buffer zone around the FR boundary.

Source: authors' construction. The map is created through QGIS software.

Social structure is thus unlikely to be a concern in our context for two reasons. First, prior work in history and anthropology clearly suggests that, while segmentary lineage was prevalent among frontier tribes, it was a less predictable force for collective organization. Tribes could organize differently at different times and 'demonstrated an ability to coalesce and dissolve in a way that was extremely difficult to predict' (Beattie 2013: 24). In this regard, Appendix A offers further details using specific illustrations and quotes. Second, tribes on both sides of the FR border predominantly belong to the same ethnic and religious group. Figure 5 compares sub-district (i.e. tehsil) level data on ethnicity and religion on both sides of the FR border. Panel A of the figure shows the percentage of the population that belongs to the major ethnicity in our area of study (Pashtuns). For most of the sub-districts either side of the FR border, between 75 and 100 per cent of the population belong to the major ethnic group. The remaining sub-districts are also quite evenly balanced in terms of their ethnic make-up, with the difference in the percentage of Pashtuns on either side of the border exceeding 25 percentage points in only a handful of cases. In panel B we show the almost perfect balance in the religious make-up of the population on either side of the FR border. Remarkably, for each of the sub-districts on either side of the FR border, the percentage of the population that belongs to the majority religion (Islam) is always between 75 and 100 per cent.

Figure 5: Balance across the FR border in ethnicity and religion

Panel A: Percentage of Pashtuns at sub-district level



Panel B: Percentage of Muslims at sub-district level



Note: this figure shows the distribution of the majority ethnic and religious groups on either side of the FR border at the sub-district level. Panel A shows the balance between the majority ethnic group (Pashtuns) across the border, whereas panel B shows the balance for the majority religious group (Muslims).

Source: authors' compilation.

## 4.3 Baseline estimates

We now present the SRD estimates of the relationship between frontier rule and the incidence of conflict against the state. We first examine the raw relationships in our SRD sample using three measures of conflict against the state: number of incidents, number of deaths, and number of injuries. Figure 6 shows the binscatter plots of the unconditional relationship between each of the three measures of conflict against the state and distance from the FR border. Following the best practices of visual inference proposed in Korting et al (2023), we use small bins (19 bins of size 5km each) and default y-axis scaling.<sup>34</sup> As the three plots in Figure 6 show, even in the raw data, a strong discontinuity in conflict is visible at the FR border. In Appendix B, Figure B3, we show pretty much the same pattern exists for smaller bins (of size 2.5km each). Specifically, moving from just outside to just inside the FR border, there is a clear discontinuous increase in conflict against the state.

Additionally, in Appendix B, Table B1, we extend our analysis and re-estimate our baseline SRD specification represented by Equation 1 for a broader measure of violence against the state that includes targets such as police, government officials, roads, bridges, etc., in addition to the military. As alluded in Section 3.1, details of this broader measure of state-directed violence are given in Appendix A. What Table B1 shows is that even if we were to use this broader measure of violence against the state that includes a whole range of state-affiliated institutions and infrastructure, our results essentially remain unchanged. Whether we consider models with controls (columns 2, 4, and 6) or without (columns 1, 3, and 5), the estimates show that FR still leads to a discontinuous rise in violence against the state. Figure B4 in Appendix B shows the binscatter plots associated with the broader measure, and even here the rise in state-directed violence just as one moves inside the FR border is quite clear.<sup>35</sup>

# 4.4 Robustness checks

We now conduct a battery of robustness checks to test the sensitivity of our baseline results in Section 4.3. We begin by showing in Table 5 the robustness of our baseline estimates to restricting the sample to grid cells that lie within two alternative buffer zones around the FR border—one broader at 60 km from the border and the other narrower at 40 km from the border. In Panel A of Table 5, we report estimates from the sample within the 60-km buffer zone. Panel B reports results for the sample restricted to a 40-km buffer zone on both sides of the FR border. As the results show, regardless of these restrictions imposed on the sample, the effect on conflict of moving from just outside to just inside the FR border is both positive and statistically significant. Furthermore, the estimated SRD effect is consistently positive and statistically significant in both models with controls (columns 2, 4, and 6) and without controls (columns 1, 3, and 5).

In Table 6 we test for the sensitivity of our baseline estimates to the inclusion of fixed effects for shorter border segments (i.e. 18 km and 15 km instead of 20 km). As border segment fixed effects account for treatment effect heterogeneity along the FR border, this sensitivity analysis is important to the robustness of our findings. Whether we restrict the length of the border segments to 18 km (Panel A) or 15 km (Panel B), we find a positive and statistically significant effect of frontier rule on conflict against the state. Next, we replicate our results using an alternative source

<sup>&</sup>lt;sup>34</sup> As Korting et al. (2023) show, the size and spacing of bins along with other recommended conditions perform better on several econometric criteria. Selection of smaller bins, in particular, leads to lower type-I error rates.

<sup>&</sup>lt;sup>35</sup> The visual discontinuities using the broader measure in Figure B4 appear somewhat localized around the actual FR border itself. However, this is clearly not the case for our preferred narrower measure that only focuses on military attacks (see Figure 6).

for conflict data. Our baseline estimates are based on conflict events recorded in the GTD (2021). A potential concern is the existence of measurement error in the recording of conflict events, especially in the earlier time period covered by the GTD. This is unlikely to be a serious concern in our case. The underlying data is based on reports from a variety of open media sources that have been independently verified as being credible. Also, the frontier areas of Pakistan are some of the most intensely studied border areas in the world by historians, anthropologists, and international relations scholars. A systematic omission of conflict events is thus highly improbable. Nevertheless, to allay potential concerns about the source of the data, we re-estimate the baseline specification using an alternative source of geocoded information on conflict events: the Uppsala Conflict Dataset (UCD) (UCD 2023). The main advantage of the UCD is that it is the 'oldest ongoing data collection project for civil war, with a history of almost 40 years'.<sup>36</sup> Unlike the GTD, however, the UCD does not disaggregate conflict events by target types, which means that we cannot create like-for-like measures of conflict between the two data sources.

Figure 6: Conflict against the state and distance to FR border (the raw relationship)



(a) Number of incidents

<sup>&</sup>lt;sup>36</sup> The link to Uppsala Conflict Dataset (UCD) website is here: https://www.pcr.uu.se/research/ucdp/about-ucdp/.

#### (b) Number of deaths in incidents



(c) Number of injuries in incidents



Note: binned scatterplots (19 bins of size 5km each) of the unconditional relationship between conflict against the state and distance to the FR border. The y-axis reports the natural log of 1 plus the incidence of conflict against the state for each of our three measures. The x-axis reports the distance (in km) from the FR border for areas under FR and non-FR. The border itself is at km 0 with positive values indicating km inside the FR territory.

Source: authors' construction.

Notwithstanding this limitation, in Table 7 we present the SRD estimates of frontier rule on the two measures of *overall* conflict (number of conflict incidents and number of deaths in conflict incidents) for which information is available in the UCD. For both measures, and regardless of whether controls are included in the specification (columns 2 and 4) or not (columns 1 and 3), our baseline results are strongly validated by the UCD data.

Another concern is that our results might be capturing the effect of a *border* rather than the effect of being inside the border that emanates from an institutional discontinuity. Such a concern is important to address as recent literature demonstrates that violence is systematically higher in closer proximity to the border (Michalopoulos and Papaioannou 2016; Depetris-Chauvin and Özak 2020). A related concern is that our SRD estimates could be potentially susceptible to some ambiguity in the treatment status. This could happen if, for instance, grid cells very close to the border have some of their area outside FR (and are therefore not treated) and some inside FR (and thus treated). Including such grid cells in our sample would bias our estimates. To rule out both concerns, we exclude all grid-cells from our sample that are close to the FR border. Following standard practice in the literature, we conduct donut-hole analysis whereby grid cells within 0.5 km of the FR border in either direction are excluded. As Table 8 shows, our results for all three measures of conflict against the state continue to hold despite this exclusion. As a further robustness exercise, we successively reduce the radius of donut hole even further from 0.5km to 0.1km. The resulting SRD estimates and their confidence intervals for the main measure based on conflict incidents are visually displayed in Figure 7. As the figure shows, despite the successive deletion of contiguous grid cells, our regression discontinuity (RD) estimates remain positive and statistically significant (see Figure B5 in Appendix B for similar plots for the other two conflict measures).



Figure 7: SRD estimates after excluding spatial units close to the FR border

Note: this figure shows estimated coefficients and confidence intervals for SRD specifications run on different restricted samples that successively exclude observations very close to the FR border. For example, 'donut\_1' excludes observations that are within 0.5km of the border, 'donut\_2' excludes observations within 0.4km of the border, etc.

Source: author's construction.

Next, we conduct several additional robustness tests and report the results in Tables B2–B5 of Appendix B. To assess the robustness of our findings to alternative functional forms, we first increase the order of the polynomial of our running variable (distance to the FR border). As Cattaneo et al. (2019) argue, this can reduce the approximation error in estimating the RD effect. Accordingly, we use a 'quadratic' polynomial as the functional form for the running variable in Table B2 and show that our estimates remain consistent both in terms of sign and statistical significance. Next, in Table B3, we report the SRD estimates that use latitude, longitude, and their interactions as the running variables rather than the Euclidean distance. An advantage of this approach is that it helps to account more directly for features that vary over a two-dimensional space (Dell 2018; Moscona et al. 2020). We report estimates for specifications with a linear (Panel A) and quadratic polynomial (Panel B) in latitude and longitude. Following Dell (2018), we include fixed effects for four broadly defined segments of the frontier rule border and use Conley standard errors to account for spatial correlation. As Table B3 shows, the main results remain unchanged.

As another robustness exercise, we choose a manual approach towards bandwidth selection as opposed to the data-driven approach employed in our baseline estimation. Reassuringly, whether we impose a bandwidth of 15, 12, or 10 km on either side of the FR border, our SRD estimates remain consistent both in terms of sign and statistical significance (see Table B4). We also test the robustness of our findings to the use of a different kernel weighting strategy for observations close to the FR border. Rather than using a triangular kernel, which assigns a *linear* decaying weight to observations as one gets further away from the border cut-off, we use an Epanechnikov kernel, which gives a *quadratic* decaying weight.<sup>37</sup> Reassuringly, the results are strongly consistent with our baseline estimates (see Table B5).

Finally, we address the potential concern that, rather than capturing the genuine institutional effect of the border demarcating frontier rule, our results could simply be reflecting structural differences in the northeast–southwest dimension. Accordingly, in Table B6 we apply a falsification test that consists in shifting the original FR border south-westward and seeing whether our results still hold up (see Figure B6 in Appendix B for a map showing both the original and the shifted FR borders). As the table shows, there are no statistically significant differences in violence against the state between grid cells on either side of this placebo border, thereby confirming that our results cannot be attributed to structural characteristics that vary in a northeast–southwest dimension.

## 5 Mechanisms

We have thus far shown that FR areas were more predisposed to sovereignty-contesting forms of violence in the long run. In this section we furnish additional evidence, both quantitative and qualitative, to elaborate our causal narrative and the proposed mechanism. We begin with an assertion repeatedly made in the qualitative literature that the post-9/11 landscape is crucial to understanding the rise of violence in Pakistan's north-western frontier (Tripodi 2009; Nawaz 2011; Naseemullah 2014). Our results lend credence to this. Utilizing the temporal dimension of our dataset, we first show that the difference in conflict incidence between FR and non-FR areas only emerged after 9/11 when the military moved into frontier regions on US insistence. It is important to note that the Pakistani military initially moved *only* to guard the border with Afghanistan and

<sup>&</sup>lt;sup>37</sup> Both kernels assign zero weight to observations that are strictly outside the bandwidth over which the SRD estimates are computed.

prevent possible infiltration. In military jargon, this was an effort to 'dominate the space' without carrying out any operations. However, as Nawaz (2011: 6) argues:

when the Pakistan Army rolled into FATA after 9/11, they faced a wall of resentment as an 'alien force' doing the bidding of a foreign power, the United States.

The extension of coercive authority in frontier regions, along with the Pakistani state's decision to join the US-led war on Afghanistan, was deeply unpopular. This led to a sudden rise in local grievances and exposed the institutional fragility of frontier rule. As argued before, social order in frontier regions was maintained through an elite-negotiated rule, which is more negotiable and fluid in character. Three features of this rule made it particularly vulnerable to the 9/11 shock: (a) absence of formal institutions of conflict management; (b) overwhelming reliance on local elites; and (c) low trust in the state. We furnish concrete evidence on all three dimensions. Furthermore, we show that as a local insurgency took hold around 2004, tribal elites became an important casualty. Their strategic targeting and assassination effectively removed the main pillar of frontier rule, thereby intensifying conflict and unravelling social order. Our evidence is consistent with the prediction in Naseemullah (2014: 507) that when frontier rule is 'marginalized or disrupted [...] channels of communication are weakened and commitments become less credible. Consequently, tensions are much more likely to explode into large-scale social disorder'.

To foreground our analysis, Figure 8 provides a timeline of major developments in frontier areas in the post-9/11 period. These include the endogenous military response to local insurgency in the guise of counterinsurgency operations conducted by the Pakistani military and selected US drone strikes, both of which intensified after 2009 and further escalated violence in a mutually reinforcing manner. This is consistent with prior evidence in the literature showing that the use of 'indiscriminate violence' and 'overwhelming fire power' can be counter-effective and can intensify insurgency (Kocher et al. 2011; Cederman and Vogt 2017; Dell and Querubin 2018). As the foregoing discussion will show, the timing and sequencing of these events form a crucial part of our causal narrative. We organize our discussion in three inter-related steps. First, we parse our analysis into the pre- and post-9/11 periods and show that our headline finding is driven by the latter period. Second, we furnish evidence on the institutional fragility of frontier rule and the targeting of tribal leaders. Finally, we rule out potential competing explanations for our results.



Figure 8: A timeline of major developments in the frontier region in the post-9/11 period

Source: authors' construction based on secondary literature.

#### 5.1 The 9/11 shock and conflict in FR areas

We begin by visually inspecting the trajectory of conflict incidents against the state, measured on a per capita basis and averaged over five-year windows during the period 1971–2018. We chart the evolution of these conflict incidents separately for FR and non-FR regions in an unrestricted sample covering all sub-districts (*tehsils*) of Pakistan. As expected, Figure 9 shows no noticeable difference in the evolution of conflict against the state between FR and non-FR areas in the period preceding 2001. However, the conflict trajectory of FR regions begins to diverge from that of non-FR areas after 2001. There is a noticeable rise in violent attacks against the state in FR areas from 2001 onwards. After a steady increase till 2010, there is a steep rise in violence during the 2011–15 period. The attacks in FR areas then begin to subside after an important phase of military operations between 2014 and 2017 (code-named *Zarb-e-Azab*), which represented the state's most concerted response to the local insurgency. However, there is still a discernible difference in conflict incidence between FR and non-FR regions in the 2016–18 period, indicating that by no means did the violence revert to its pre-2001 levels.



Figure 9: The evolution of conflict incidents in Pakistan, pre- and post-9/11

Note: this figure charts the evolution of conflict incidents between the FR and non-FR areas averaged over fiveyear windows during the period 1971–2018. The unit of observation is the sub-district (i.e. *tehsil*). The variable on the y-axis is the number of conflict incidents against the state per capita.

Source: authors' construction using GTD (2021).

To formally test the salience of the 9/11 shock for our causal narrative, we separately re-estimate the SRD specification for the pre- and post–2001 period. As before, the unit of observation is a 10km-by-10km grid cell, and the sample is restricted to a buffer zone of 50 km around the FR border. The results, reported in Table 9, reveal a clear pattern. There is no statistically significant discontinuity in the pre-9/11 period in violent conflict against the state, whether measured by number of conflict incidents against the state (columns 1–2), number of deaths in incidents against the state (columns 3–4), or number of injuries in incidents against the state (columns 5–6). The discontinuity emerges only after 9/11, when areas just inside the FR border witnessed a significantly higher level of conflict against the state than areas just outside it, as manifested in consistently positive and statistically significant coefficient estimates in columns 7–12. The results

also hold when we consider conflict against the state within a wider 60-km buffer zone around the FR border (see Appendix B, Table B7).

# 5.2 Institutional fragility of frontier rule

We next present evidence on the three inter-related features of frontier rule that made it vulnerable to the 9/11 shock: (a) the significantly lower reliance of frontier residents on formal institutions of conflict management (b) the greater role of tribal elites in local dispute resolution, and (c) lower trust in the formal institutions of the state. We utilize data from a nationally representative survey of individuals carried out by the Free and Fair Election Network (FAFEN) in 2016 and estimate linear probability models to probe whether residents in FR areas had less reliance on elected members of parliament and greater recourse to tribal *jirga* (consultative assembly of elders). As argued in Section 2, frontier residents remained practically disenfranchised till 1997, when legislators from the region were directly elected to the National Assembly of Pakistan Nevertheless, adult enfranchisement remained limited as elections were held on a non-party basis and political parties were not allowed to function in these regions. Consequently, tribal *Maliks* maintained their de facto control over all aspects of local society (Anwar and Cheema 2017; Ullah and Hayat 2017).

We construct three binary dependent variables: (a) whether an individual had recourse to a member of the national assembly (MNA) for dispute resolution; (b) whether the individual had no contact with an MNA in the last two months; and (c) whether the individual had recourse to a tribal assembly (*jirga*). Our main explanatory variable is a dummy variable that is equal to one when the surveyed individual resides in an FR area. The control variables, when included, are age (in natural log), locality-level fixed effects, and dummy variables for gender, educational status, source of income, and household income range. The results are reported in Table 10. The evidence is consistent with our priors. Surveyed individuals in FR areas had limited recourse to their elected representatives (i.e. MNAs) for dispute resolution (columns 1–2). They were also systematically less likely to have established contact with the MNA during the previous two months at the time of the survey (columns 3–4). Importantly, FR residents were significantly more likely to resort to the tribal *jirga* for dispute resolution (columns 5–6). The FR status dummy, which picks out individual respondents residing in FR areas, is statistically significant in all specifications including those with controls (columns 2, 4, and 6).

Overall, this evidence is consistent with findings from other independent surveys that highlight the significance of *jirga* as the principal dispute resolution mechanism in FR regions. For example, findings from a detailed survey conducted in tribal areas showed that nearly three quarters (74.40 per cent) of respondents were aware of the *jirga* (Shinwari 2011: 66). As Shinwari (2011: 66) argues, this 'high level of awareness is due to the fact that the jirga is the only justice dispensing mechanism in FATA both accessible and trusted by many'. Further evidence of the high level of trust in traditional modes of dispute resolution can be gleaned by the fact that the majority of respondents (91 per cent) were satisfied with the delivery of justice through the *jirga* (Shinwari 2011: 81). When asked in the same survey in 2011, tribal elders were considered as the most trusted institution (20 per cent). By contrast, elected representatives (i.e. MNAs) received the lowest rating on trust (1 per cent) (see Shinwari 2012: 50). Survey evidence also supports the claim that tribal leaders serve as the main link between local populations and the state. About 43 per cent of survey respondents had approached the *jirga* to resolve their disputes; another 31 per cent initially took their disputes to local leaders (*Khans* or *Maliks*), while only 6 per cent of respondents resorted to courts in adjoining districts of Khyber Pakhtunkhwa (Shinwari 2011: 80).<sup>38</sup>

To formally investigate whether FR residents have systematically lower levels of trust in state institutions, we again use the 2016 FAFEN survey and report the results for linear probability models in Table 11 where the dependent variable is a binary indicator that is equal to one when a respondent expresses 'very little or no trust' in state institutions. As before, our main explanatory variable is a dummy for whether an individual resides in a household that lies inside the FR boundary. For each indicator we report estimates with and without controls (described earlier). The results confirm our priors and are consistent with the empirical patterns presented so far. As is evident from Table 11, the coefficients on the FR status dummy are consistently positive, statistically significant at the 1 per cent level, and robust to the inclusion of controls. Residents in FR areas are significantly more likely to have 'very little or no trust' in the parliament (columns 1– 2), the district court (columns 3–4), the high court (columns 5–6), and the supreme court (columns 7–8).

# 5.3 Assassinations of tribal leaders

We have so far demonstrated that the primary (and preferred) mode for conflict resolution in FR areas remains the *jirga* system that is overseen by tribal elders. We have also shown that FR residents have limited access to formal institutions of conflict management and exhibit less faith in state institutions. We next show that the disruption to frontier rule in the wake of the 9/11– induced shock primarily took place through the systematic and strategic targeting of tribal elders.<sup>39</sup> To demonstrate this we return to our SRD design which uses 10km-by-10km grid cell-level data on conflict incidence and restricts the sample within a 50-km buffer zone around the FR border. As our main focus here is on explaining post–9/11 violence, we also restrict the sample to the period after 2001. However, we now replace our dependent variable with the number of attacks against tribal elders, the number of deaths in attacks against tribal elders, and the number of injuries in attacks against tribal elders, respectively. The overall empirical set-up remains the same as before.

The results, reported in Table 12, reveal a statistically significant discontinuity whereby regions just inside the FR border witnessed a systematically higher number of attacks against tribal elders than areas just outside the FR border (columns 1–2). The same pattern holds for the number of deaths in attacks against tribal elders (columns 3–4) and the number of injuries in attacks against tribal elders (columns 5–6). These results suggest a discontinuous rise in targeted attacks against tribal elites in the post-9/11 period as we move from just outside to just inside the FR border. Next, we provide a specific test of whether such a strategic targeting of tribal elders led to a significant escalation of violence. Specifically, we ask whether spatial units in the FR region that witnessed an attack on a tribal elder for the first time experienced more intensified conflict in subsequent years relative to spatial units that did not witness such an attack. To investigate whether the first-ever attack on a tribal elite preceded the intensification of conflict, we estimate a version of the difference-in-differences regression specification that uses leads and lags to estimate the effects of before, during, and after a tribal elder is first targeted:

<sup>&</sup>lt;sup>38</sup> As per rules, residents of FR regions can appeal in the court of the commissioner of the adjoining district against the verdict of the *jirga*.

<sup>&</sup>lt;sup>39</sup> Figure B7 in Appendix B provides a clear spatial discontinuity in the targeted assassinations of tribal elders in the post-9/11 period. More attacks against tribal elders are observed in the FR region relative to the non-FR region.

$$y_{i,s,t} = v_s + \gamma_t + \sum_{\tau=0}^m \delta_{-\tau} * D_{i,s,t-\tau} + \sum_{\tau=1}^q \delta_{+\tau} * D_{i,s,t+\tau} + \epsilon_{i,s,t}$$
(2)

where  $y_{i,s,t}$  is a measure of overall conflict incidence (i.e.  $\ln (1 + all \text{ conflict incidents})$ ) in grid cell *i* within district *s* in year *t*.  $D_{i,s,t}$  is an indicator for when an attack against a tribal elder takes place for the first time in grid cell *i* in district *s*.  $v_s$  are district fixed effects that control for time-invariant characteristics at the district level.  $\gamma_t$  are year fixed effects that control for shocks common to all districts in any given year.  $\sum_{\tau=0}^{m} \delta_{-\tau} * D_{i,s,t-\tau}$  allow for *m* lags  $(\delta_{-1}, \delta_{-2}, ..., \delta_{-m})$  or 'post-tribal elders attack effects' and  $\sum_{\tau=1}^{q} \delta_{+\tau} * D_{i,s,t+\tau}$  allow for *q* leads  $(\delta_{+1}, \delta_{+2}, ..., \delta_{+q})$  or 'pre-tribal elders attack effects'. The former is referred to as 'post-treatment effects' and the later as 'anticipatory effects' in the standard treatment effects literature.<sup>40</sup> The estimated leads and lags, with respective coefficients and confidence intervals, are plotted in Figure 10. The estimated effects reveal an interesting pattern. Although there is a gradual uptick in violence several years before a tribal elder is first attacked in a grid cell, there is a noticeable jump in violence after such an attack. The sharply increasing effects persist even ten years after the attack first took place. This clearly suggests that targeted attacks on tribal elders leders leders a subsequent escalation of violence.

Taken together, these results suggest that the strategic elimination of tribal elders removed the main pillar of social order and, in the absence of any formal avenues for conflict management, created an institutional void that served as a fertile breeding ground for conflict against the state. Our evidence is consistent with both factual and qualitative research accounts, including the argument made by Naseemullah (2014: 515) that 'the disruption of the frontier rule that lay at the heart of political order in FATA is responsible for the nature and extent of conflict following the exogenous shock of the war'. It has been noted that attacks on tribal leaders resulted in more than 90 per cent of tribal elders being either killed or wounded in FR areas (GTD 2021; SATP 2021). Around 150 *Maliks* were killed in 2008 alone (Fair and Jones 2009; Zahab 2013). As Naseemullah (2014: 518) argues, 'this struck a serious blow against the structures within tribal society that were successful interlocutors with the state' and helped to maintain political order.

<sup>&</sup>lt;sup>40</sup> See Chapter 5 of Angrist and Pischke (2009).



Figure 10: The estimated impact of targeted attacks on tribal elders on conflict escalation

Note: this figure plots the coefficients and confidence intervals for the estimated relationship between the first ever attack on a tribal elder and the overall level of conflict in the same grid cell. The dependent variable is  $\ln (1 + all conflict incidents)$ . Estimates are from the model in Equation 2 that allows for effects before, during and after the first ever time a tribal elder is attacked and includes district and year fixed effects.

Source: authors' construction using GTD (2021).

## 5.4 Competing explanations

Thus far we have attributed the discontinuous rise of violence against the state in the post-9/11 period to the in-built vulnerability of frontier rule to external shocks. In this section, we rule out some important competing explanations for our empirical findings. Specifically, we try to address four alternative explanations built around: (a) the potential conflict spillover from Afghanistan; (b) income shocks; (c) US drone strikes; and (c) pre-9/11 differences in economic development.

## Spillover from Afghanistan

A potential concern is that the relative uptick in violence in FR areas after 9/11 may have been caused by the spillover of conflict from Afghanistan rather than the fragility of underlying institutional arrangements. Such spillover is possible for several reasons. Afghanistan shares a long border with Pakistan (1,640 miles), a sizeable portion of which is with frontier areas (i.e. 373 miles). Following the US invasion in 2001, Afghanistan witnessed an active phase of conflict. Given the presence of Taliban sympathizers in Pakistan, this conflict might have spilled over into FR areas. Pakistan has also historically hosted millions of Afghan refugees since the Soviet invasion in December 1979. While these refugees settled in different parts of Pakistan, including major urban centres such as Lahore and Karachi, and were predominantly civilians fleeing from war and instability, it is possible that some militants may have retreated into north-western Pakistan in the guise of refugees and launched attacks against state installations in Pakistan as a retribution for the country's support for the US-led war.

While it is difficult to completely rule out the conflict spillover from Afghanistan, we present several pieces of evidence to suggest that the post-9/11 disruption of frontier rule in Pakistan

cannot be entirely attributed to such spillover. We begin by showing in Figure 11 the spatial distribution of conflict incidents in Afghanistan and Pakistan. As Figure 11 shows, there is a high concentration of attacks against the state on the Afghan side of the Durand line, the international border separating Afghanistan and Pakistan. It is important to highlight that the Afghan border, especially the segment adjacent to FR area, has been tightly monitored and surveilled in the post-9/11 period. Soon after 2001 attack on Afghanistan, the US sponsored the Border Security Program to secure the FR border through strict aerial and on-the-ground oversight. It entailed the construction of 137 border outposts and provision of surveillance, it is still possible that some militants from Afghanistan might have slipped through the border into Pakistan's frontier areas and instigated violent actions against the state.

In what follows, we provide an array of evidence to argue that this possibility, although remote, is unlikely to systematically explain our results. Firstly, to rule out spatial spillover from Afghanistan, we compute the distance of the centroid of each grid cell from the Durand line border and include it as an additional covariate in our SRD specification. As Table 13 shows, the results survive. In a second related exercise, we exclude all grid cells in our 50km buffer zone that are contiguous to the Afghan border. These grid cells can be most worrisome in terms of potential spillover. As Table 14 reveals, our results remain unchanged on this restricted sample. As a third strategy, we utilize the information in the GTD to identify the perpetrators or originators of attacks against state targets. In Table 15 we categorize these attacks by type of militant outfits, distinguishing between local, foreign, and unknown organizations. Information is provided for the whole sample period, 1970-2018 (columns 1-2), the pre-2001 period (columns 3-4), and the post-9/11 era (columns 5-6). Regardless of whether we restrict this exercise to the whole period or to the post-9/11 era, more than 95 per cent of the known attacks were claimed by local outfits rather than Afghan-based militants.<sup>42</sup> This includes Pakistan-based outfits, such as the Tehreek-i-Taliban Pakistan (TTP), Lashkar-e-Jhangvi, Tehreek-i-Nifaz-e-Shariat-e-Mohammadi, and Sipah-e-Sahaba, among others.

These results are consistent with extensive qualitative literature highlighting the local roots of the insurgency in FR areas (Jones and Fair 2010; Nawaz 2011; Naseemullah 2014; Watts et al. 2014). There are essentially two main reasons why Afghan militants may have desisted from launching widespread attacks against the Pakistani state in FR regions. First, it is generally difficult for cross-border militants to develop and sustain an organized network in another country. In addition to requiring grassroots support from the local populace, they need a clandestine military architecture which keeps fighters supplied with both weapons and rations. While there is plenty of evidence that local insurgent groups like the TTP developed such an architecture in Pakistan's tribal areas along the Afghan border, the same is not true for Afghan groups (Elahi 2019). Instead, the Afghan factions were more likely to have used Pakistan's FR areas as a sanctuary from where they could launch cross-border attacks against US forces inside Afghanistan.

<sup>&</sup>lt;sup>41</sup> For further details, see the report submitted to the US Government Accountability Office (GAO), report number GAO-09-263SP entitled *Securing, Stabilizing, and Developing Pakistan's Border Area with Afghanistan: Key Issues for Congressional Oversight.* Available at: https://www.gao.gov/assets/a286308.html.

<sup>&</sup>lt;sup>42</sup> Table B8 in Appendix B provides details of attacks by various militant groups, offering a broader definition of 'sovereignty-contesting' forms of violence. This table encompasses attacks against law enforcement personnel (including military, police, and their sub-branches), educational institutions, healthcare facilities, politicians, civil servants (such as teachers, doctors, lawyers, judges), diplomats, roads, bridges, railroads, airports, utilities installations, and more.

The second reason is related to the Pakistani state's nuanced strategy for dealing with militant outfits within its borders. In the wake of 9/11 the deep state, primarily comprising Pakistan's military and intelligence services, 'established a differentiated framework for dealing with divergent outfits' (Lynch III 2018: 68). As part of this, any group that 'remained supportive or neutral in its approach to the Pakistani state' would often be overlooked and left to its own devices (Lynch III 2018: 68). By contrast, groups that 'threatened the Pakistan state or viewed international Islamist jihad as the highest order priority' would be dealt with severely (Fair et al. 2010; Hussain 2005; Rana 2004). Faced with this differentiated approach, Afghan groups were less likely to engage in systemic violence against the Pakistani state.



Figure 11: Post-9/11 conflict incidents in Afghanistan and Pakistan

Note: this figure shows the distribution of conflict incidents in Afghanistan and Pakistan. The dark red line represents the Durand line, the international border between Afghanistan and Pakistan. Each dot represents an attack against the state. The black dots (on the right of border) are attacks against the state in Pakistan whereas the red dots on the left of the border denote similar attacks in Afghanistan.

Source: authors' construction using GTD (2021).

Even if we were to admit the possibility of a cross-border spillover of violence, it is not easy to understand why the Afghan attackers would stop at the FR border and not engage in higher-profile targets in settled regions. Indeed, after Musharraf's decision to join the US-led war in Afghanistan, major Pakistani cities (e.g., Peshawar, Lahore, Rawalpindi, Karachi) witnessed a spate of violent attacks which became important national incidents and key pressure points for public policy. Finally, if conflict did indeed spill over from Afghanistan to FR areas, why do we not see a similar spillover in the lower half of the Afghanistan-Pakistan border (see Figure 11)? Taken together, both the quantitative and contextual evidence reassure us that the potential overflow of conflict from Afghanistan was not the main driver of the post-9/11 trajectory of violence against the state in FR areas.

#### Income shocks

Another concern is that the post-9/11 spike in conflict in frontier areas may have been caused by a possible disruption to local economy after 2001 rather than the underlying fragility of frontier rule. This is an important concern as we know from prior work that income shocks are an

important driver of conflict (Blattman and Miguel 2010). The main economic shock that frontier areas witnessed after 2001 was the disruption of local economic activity induced by military operations (Jones and Fair 2010; Nawaz 2011). Even if concentrated and targeted, these military operations did to some extent displace populations, disrupt local economic activities, and therefore represented a shock to incomes. However, we can immediately rule out such a concern by showing that the rise of local insurgency in the FR areas pre-dates the military operations, which were in fact an endogenous response to the uptick in violence.

To do so, we estimate a basic event study specification that charts the differential evolution of conflict incidence between FR and non-FR areas before and after 2001:

$$y_{i,d,t} = v_d + \gamma_t + \Gamma_t(Frontier_Rule_{i,d} \times Year_t) + \epsilon_{i,d,t}$$
(3)

Here,  $y_{i,d,t}$  is a measure of overall conflict incidence (i.e.  $\ln(1 + all \text{ conflict incidents})$ ) in subdistrict *i* within district *d* in year *t*.  $v_d$  and  $\gamma_t$  are district and year fixed effects, respectively. The term (*Frontier\_Rule*<sub>*i,d*</sub> x *Year*<sub>*t*</sub>) is the interaction between the indicator for whether sub-district *i* is exposed to historic frontier rule and the full set of year fixed effects. Finally, the vector of estimated interaction coefficients,  $\Gamma_t$ , shows the relationship between frontier rule and conflict over time. Specifically, it charts the evolution of conflict in FR areas relative to non-FR areas in each year of our conflict panel from 1970 to 2018.<sup>43</sup>

The main result for the specification in Equation 3 is displayed in Figure 12, which plots the point estimates and corresponding confidence intervals on the dummy for frontier rule interacted with year fixed effects. As the figure shows, the conflict trajectory of FR areas begins to diverge from non-FR areas soon after 2001. There is a clear increase in the coefficient of interest in 2004. This coincides with the year when local insurgency begins to gain momentum. The coefficient on FR interaction dummy progressively increases until it becomes statistically significant in 2008 and reaches its peak in 2012. This is the period during which a concerted endogenous response to local insurgency appears via targeted military operations and drone strikes. Importantly, this period also witnessed the highest number of attacks against tribal elders, which forms a key part of our mechanism (see Section 5.3). Finally, there is a noticeable drop in the coefficient capturing the differential FR effect in 2014. This coincides with the launch of the *Zarb-e-Azb* in 2014, the most serious and effective military operation to date (see Section 2.2). However, even if the FR interaction coefficient declines after 2014, it remains considerably higher in magnitude relative to the pre-2001 period, suggesting that by no means does the overall conflict incidence in FR areas revert to its pre-9/11 levels even after the 2014 military operation.

<sup>&</sup>lt;sup>43</sup> The estimated interaction coefficients in  $\Gamma_t$  have to be measured relative to a base time period, which we take to be the first year in our conflict panel: 1970.
Figure 12: The frontier rule effect over time



Note: this figure plots the point estimates and confidence intervals of the frontier rule (FR) dummy interacted with year fixed effects for the specification in Equation 3. The coefficient for the base time period (1970) was set to zero and is not shown in the figure. The dependent variable is the log of (1 + all conflict incidents). The FR dummy takes the value of 1 for areas that are exposed to historic frontier rule. Analysis is carried out at the sub-district (*tehsil*) level, and district and year fixed effects are controlled for. Sample includes all sub-districts of Pakistan.

Source: authors' construction based on data from GTD (2021).

Taken together, the timeline in Figure 8 and the event study specification plot in Figure 12 clearly show that the uptick in violence in the FR areas (in 2004) pre-dates the intensive phase of military operations (2009–14), thereby offering conclusive evidence against income shocks being the primary explanation for our results. What these results also highlight is the salience of timing and sequencing of key events in our causal story. This closely follows the spirit of Pierson (2004: 67) who argued that 'long-term outcomes of interest depend on the relative timing of important processes [...] A variable's impact cannot be predicted without an appreciation for when it appears within a sequence unfolding over time'. This is especially important in the context of insurgency-based violence where several mutually reinforcing factors are triggered in a sequence that is drawn out over time.

# Drone strikes

An important part of the US war on terror was the use of unmanned drones to take out specific insurgent targets in Afghanistan and north-west Pakistan, including frontier areas which are the focus of this study. While drone strikes specifically targeted the militants, their collateral damage sometimes included the deaths of innocent civilians. This might have led to a rise in local grievances. In a survey in frontier areas, 63 per cent of respondents considered drone attacks as

'never justified' (Shinwari 2012: xvi and 88).<sup>44</sup> Such grievances can trigger further conflict. Recent studies offer mixed evidence on the possible impact of drone strikes on violence, with some authors establishing a decrease in militant violence in the wake of drone strikes (Johnston and Sarbahi 2016) and others documenting a positive impact of such strikes on terrorist violence (Mahmood and Jetter 2019). The latter use a novel instrumental variable strategy to show that drone strikes increased anti-US sentiment amongst both insiders (members of terrorist organizations) and outsiders (the Pakistani populace), thereby translating into greater violence. Other more general work in the area has argued that drone strikes could heighten local grievances and provide ideological fodder to local insurgencies (Hudson et al. 2011).

This raises the possibility that the post-9/11 rise in state-directed violence in FR areas can potentially be a result of unpopular drone strikes rather than, as we argue, the institutional fragility of frontier rule in the face of external shocks. It is easy to rule out such a concern. As we argued before, drone strikes were an endogenous response to the rise in localized violence that had already begun in 2004 (see Figures 8 and 12). As such, these strikes might have fed into the ongoing conflict. But they were not the cause of the original rise in violence in FR areas. However, to the extent that drone strikes might interfere with our explanation of the core results, we devise a novel test whereby we exclude from our sample all the grid cells that witnessed such attacks and re-estimate the SRD specification. To do so, we leverage a geo-coded dataset on the location of drone strikes compiled by Usmani (2017). The dataset provides extensive information on the 397 drone attacks in Pakistan, including those in FR regions, during the 2004–16 period.

Figure 13 provides a visual representation of these attacks in our area of study. Using the longitude and latitude of these attacks, we systematically drop those 10-by-10km grid cells where a drone attack had ever taken place from our regression sample. In Table 16 we report the estimates for this restricted sample and show that our results continue to hold despite the exclusion of drone-exposed grid cells. A residual concern is that drone attacks might displace or divert violence into proximate regions (i.e. neighbouring grid cells). However, as the empirical analysis in Johnston and Sarbahi (2016: Table 5) shows, there is no evidence of the spillover of such drone-induced violence into neighbouring areas. Overall, our evidence is consistent with the interpretation that drone attacks were not the primary driver for the original uptick in violence post 9/11.

<sup>&</sup>lt;sup>44</sup> US drone strikes typically relied on some tacit collaboration between the US and Pakistani military. To highlight the sort of grievances this gave rise to, Jones and Fair (2010: 70) cite a local religious leader as remarking that US 'attacks were carried out in the presence of the Pakistan Army; we cannot ignore our army's cooperation with foreign forces in actions that kill innocent people'.

Figure 13: Drone strikes in Pakistan's North-West Frontier



Note: this figure shows the distribution of drone attacks in Pakistan's north-western frontier. The dark red line represents the FR border separating FR and non-FR areas. Each black dot represents a drone strike.

Source: authors' construction based on data from Usmani (2017).

#### Public goods provision

Geographic peripheries have historically suffered from a more deficient provision of infrastructure. An important historical legacy of border buffer zones is the systematically lower investment in infrastructure (Popescu 2023). Such differential infrastructural provision can cumulatively set regions on different economic trajectories by shaping trade costs, inter-regional price differences, and real incomes (Donaldson 2018; Dell and Olken 2020). These, in turn, are important correlates of conflict. Accordingly, if there were systematic differences in public goods provision between FR and non-FR areas, this could be an alternative mechanism behind the frontier regions' greater proclivity towards conflict after 2001.

To consider this possibility, we examine discontinuity across the FR border in an array of contemporary and historical measures of public infrastructure, measured at the 10km-by-10km grid-cell level. The SRD estimates are reported in Table 17. We begin with contemporary measures of infrastructure. In column 1 we use the number of health sites per 10,000 persons in 2017 as the main outcome. In columns 2-3, we examine discontinuity in the length of roads and the length of waterways (in km), both in kilometres and measured in a pre-9/11 year (1992). Subsequently, we sequentially examine relevant measures of historical infrastructure and underdevelopment covering both the colonial and pre-colonial periods. In doing so, we are motivated by prior work on the long-run impact of British railroads (Donaldson 2018) and early Roman roads on contemporary infrastructure provision and economic prosperity (Dalgaard et al. 2022). Similarly, ancient trade routes are an important predictor of the spread of Islam (Michalopoulos et al. 2018), urban growth and prosperity (Blaydes and Paik 2021a, 2021b; Paik and Shahi 2023). In this spirit, we investigate in columns 4-6, respectively, potential discontinuities in railroad coverage in the British colonial era (in km), distance to major roads in the Mughal Empire (c.1556–1707), and distance to Islamic trade and pilgrimage routes (c.1300-1600). As the results in Table 17 show, there is no statistically significant discontinuity at the FR border in any of these infrastructure measures.

### 6 Conclusion

Our central argument in this paper is that exceptional institutional arrangements in frontier regions, which were typically established during colonial rule and persisted into the post-colonial era, form a more fragile basis for peace in the face of shocks. The fragility of exceptional frontier rule stems from its greater reliance on elite intermediation, lower trust in the state, and the absence of formal institutions of conflict management that can channelize grievances. We argue that, when faced with shocks, such institutional fragility is likely to manifest in 'sovereignty-contesting' forms of violence. To empirically probe this argument, we combined spatially granular data on conflict with a historical frontier that demarcated a 'rule of difference' by the British Empire in Pakistan's North-West. Our results, based on a spatial RD design, show that regions just inside the historical boundary of frontier rule experienced a significant increase in attacks against the state after 9/11 when compared with regions just outside the boundary.

We argue that 9/11 was a shock to grievances against the state caused by the Pakistani state's unpopular decision to join the US-led war on terror and the extension of coercive state authority into frontier areas on US insistence. In frontier areas that had historically suffered from an institutional void, these local grievances translated into heightened conflict. We demonstrate that the 9/11-induced disruption to frontier rule seems to have taken place through the systematic targeting of tribal leaders who were the main interlocutors between the frontier residents and the state. Their removal, especially in the absence of any alternative formal avenues of conflict resolution, intensified local conflict. We rule out multiple competing explanations behind the post-9/11 rise in violence in FR areas. These include the role of civil conflict spilling over from Afghanistan, income shocks (proxied by military operations), drone attacks, and differences in public goods provision.

Our paper has important implications for the study of conflict, especially the role of institutions in driving conflict. As Kolsky (2015: 1244) noted, 'frontiers have historically been spaces with greater potential for extreme violence'. Yet, we have limited knowledge of the factors that feed into the greater proclivity of frontier spaces to violence. Evidence presented in this paper echoes the important insight in Hopkins (2020: 2) that any study of the deep drivers of violence must contend with the legacies of how imperial powers 'defined' and 'governed' their frontier territories in the nineteenth and earlier twentieth centuries. While the legacy of frontier governance has received some scholarly attention in other disciplines—notably history, anthropology, and international relations—this paper offers the first systematic empirical enquiry of its impact on contemporary conflict. In doing so, we also contribute to the literature on conflict by showing how a *specific* historically embedded institutional arrangement shaped conflict in the face of a geopolitical shock. Our analysis underscores the importance of accounting for the interplay between domestic and external factors in driving conflict.

Not only do we offer a novel institutional explanation for the greater susceptibility of peripheries to conflict, but our results also shed light on why the extension of coercive authority in areas of limited statehood often triggers more violence. Specifically, we offer a plausible explanation as to why counter-insurgency campaigns carried out after 9/11 in Asia, Africa, and the Middle East have been met with limited success. As Comfort Ero, President of the International Crisis Group, argues, a 'military-only' response without addressing the underlying institutional deficits makes for a weak strategy.<sup>45</sup> Our results also bear relevance for understanding the rise of 'Islamic militancy'

<sup>&</sup>lt;sup>45</sup> 'Negotiating a Violent Terrain: Political Engagement in the Sahel and Somalia', Keith Griffin Lecture, Oxford Centre for Islamic Studies. 8 June 2023.

in the wake of 9/11. The rise of Boko Haram in Nigeria, Al-Shabab in Kenya, and Al Qaeda in Pakistan has been spatially concentrated in regions that were once the frontiers of global empires. Our evidence on the strategic targeting of tribal elites, a key mechanism behind the escalation of violence post 9/11, has huge relevance for understanding the spectacular growth of militancy in Africa and the Middle East. An investigative report on militancy in Africa, published by Reuters in 2021, describes such assassinations as a common pattern in the conflict playbook. In the heartland of Islamic militancy in Niger, Mali, and Burkina Faso, hundreds of village elders and community leaders were abducted or assassinated. Similar patterns of assassinations have been observed in Somalia, Nigeria, and Iraq. These local leaders typically 'settled local disputes, collected taxes, and registered births and deaths' (McAllister and Marsh 2021). Their killings created a huge power vacuum, breaking the local population's link with the state and bringing life to a grinding halt. Indeed, as a political scientist quoted in Reuters (2021), argues: 'If you want maximum disorder, you kill the chief [...] If the agenda is to replace the state, killing the village chief is just the beginning of the process'.

The relevance of frontier governance extends beyond the actual physical frontiers of countries. Frontier governmentality is about 'practice' rather than 'place'; it is a conceptual category and a 'social construct' rather than an 'objective reality' (Hopkins 2020: 14–15). The frontier signifies an institutional enclosure where the claims and authority of the state are severely limited or constrained. As the examples of native American reservations, Indian princely states, and African tribal reserves show, such enclosures can also exist in the interior. Indeed, as Hopkins (2020: 15) notes, 'for many states of the modern world, the most important and extensive frontiers demarcate their interior, rather than their exterior'. To the extent that such spaces of exceptional governance are a pervasive feature of conflict-prone states, our work has a direct bearing for state fragility, which is an emerging policy concern.

#### References

- Acemoglu, D., S. Johnson, and J.A. Robinson (2001). 'The Colonial Origins of Comparative Development: An Empirical Investigation'. *American Economic Review*, 91(5): 1369–401. https://doi.org/10.1257/aer.91.5.1369
- Acemoglu, D., and J.A. Robinson (2012). Why Nations Fail: The Origins of Power, Prosperity, and Poverty. London: Profile Books.
- Acemoglu, D., T. Reed, and J.A. Robinson (2014). 'Chiefs: Economic Development and Elite Control of Civil Society in Sierra Leone'. *Journal of Political Economy*, 122(2): 319–68. https://doi.org/10.1086/674988
- Ahmed, Z.S., B. Jenkins, and W. Iftikhar (2019). Perception of Foreign Drone Strikes by Citizens: The Context of US Drone Strikes in Pakistan'. *South Asian Survey*, 24(2): 135–57. https://doi.org/10.1177/0971523118822442
- Allen, C. (2012). Soldier Sahibs: The Men Who Made the North-west Frontier. London: John Murray.
- Angrist, J. D., and J. S. Pischke (2009). *Mostly Harmless Econometrics: An Empiricist's Companion*. Princeton: Princeton University Press. https://doi.org/10.1515/9781400829828
- Anwar, S., and A.R. Cheema (2017) 'The Future of FATA: When Reform Come Knocking'. LSE Blog, 26 October. Available at: https://blogs.lse.ac.uk/southasia/2017/10/26/the-future-of-fata-whenreforms-come-knocking/ (accessed 7 March 2023).
- Arnaut, K., C.K. Højbjerg, T. Raeymaekers, and M. Gibert (2008). 'Gouvernance et ethnographie en temps de crise'. *Politique Africaine*, 3(111): 5–21. https://doi.org/10.3917/polaf.111.0005

- Asher, S., A. Campion, D. Gollin, and P. Novosad (2022). 'The Long-run Development Impacts of Agricultural Productivity Gains: Evidence from Irrigation Canals in India'. STEG Working Paper. Centre for Economic Policy Research. Available at: https://steg.cepr.org/sites/default/files/2022-06/WP004%20AsherCampionGollinNovosad%20LongRunDevelopmentImpacts.pdf (accessed 6 March 2023).
- Aydin, A. (2012). Foreign Powers and Intervention in Armed Conflicts. Redwood City, CA: Stanford University Press. https://doi.org/10.11126/stanford/9780804782814.001.0001
- Bang, J.T., and A. Mitra (2017). 'Institutions, Information, and Commitment: The Role of Democracy in Conflict'. Defence and Peace Economics, 28(2): 165–87. https://doi.org/10.1080/10242694.2016.1142745
- Bates, R.H. (2008). When Things Fell Apart. Cambridge: Cambridge University Press. https://doi.org/10.1017/CBO9780511790713
- Bazzi, S., M. Fiszbein, and M. Gebresilasse (2020). 'Frontier Culture: The Roots and Persistence of "Rugged Individualism" in the United States'. *Econometrica*, 88(6): 2329–68. https://doi.org/10.3982/ECTA16484
- Bazzi, S., A. Brodeur, M. Fiszbein, and J. Haddad (2023). 'Frontier History and Gender Norms in the United States'. NBER Working Paper 31079. Cambridge, MA: National Bureau of Economic Research. https://doi.org/10.3386/w31079
- Beattie, H. (2013). *Imperial Frontier Tribe and State in Waziristan*. London: Routledge. https://doi.org/10.4324/9781315028347
- Becker, S. (2012). "The "Great Game": The History of an Evocative Phrase'. *Asian Affairs*, 43(1): 61–80. https://doi.org/10.1080/03068374.2012.646404
- Becker, S.O., K. Boeckh, C. Hainz, and L. Woessmann (2016). 'The Empire is Dead, Long Live the Empire' Long-Run Persistence of Trust and Corruption in the Bureaucracy'. *The Economic Journal*, 126(590): 40–74. https://doi.org/10.1111/ecoj.12220
- Bellows, J., and E. Miguel (2006). 'War and Institutions: New Evidence from Sierra Leone'. American Economic Review, 96(2): 394–99. https://doi.org/10.1257/000282806777212323
- Belmonte, A., and M. Rochlitz (2019). 'The Political Economy of Collective Memories: Evidence from Russian Politics'. *Journal of Economic Behavior & Organization*, 168: 229–50. https://doi.org/10.1016/j.jebo.2019.10.009
- Besley, T., and M. Reynal-Querol (2014). "The Legacy of Historical Conflict: Evidence from Africa'. *American Political Science Review*, 108(2): 319–36. https://doi.org/10.1017/S0003055414000161
- Besley, T., and T. Persson (2008). 'Wars and State Capacity'. Journal of the European Economic Association, 6(2–3): 522–30. https://doi.org/10.1162/JEEA.2008.6.2-3.522
- Besley, T., and T. Persson (2010). 'State Capacity, Conflict, and Development'. *Econometrica*, 78(1): 1–34. https://doi.org/10.3982/ECTA8073
- Blattman, C., and E. Miguel (2010). 'Civil War'. Journal of Economic Literature, 48(1): 3–57. https://doi.org/10.1257/jel.48.1.3
- Blaydes, L., and C. Paik (2021a). 'Trade and Political Fragmentation on the Silk Roads: The Economic Effects of Historical exchange between China and the Muslim East'. *American Journal of Political Science*, 65(1): 115–32. https://doi.org/10.1111/ajps.12541
- Blaydes, L., and C. Paik (2021b). 'Muslim Trade and City Growth before the Nineteenth Century: Comparative Urbanization in Europe, the Middle East and Central Asia'. *British Journal of Political Science*, 51(2): 845–68. https://doi.org/10.1017/S0007123419000267
- Boege, V., A. Brown, K. Clements, and A. Nolan (2008). On Hybrid Political Orders and Emerging States: State Formation in the Context of Fragility'. In Berghof Handbook Dialogue 8. Berlin: Berghof Research Centre for Constructive Conflict.

- Boone, C. (2003). *Political Topographies of the African State*. Cambridge: Cambridge University Press. https://doi.org/10.1017/CBO9780511615597
- Burke, M., S.M. Hsiang, and E. Miguel (2015). 'Climate and Conflict'. *Annual Review of Economics*, 7(1): 577–617. https://doi.org/10.1146/annurev-economics-080614-115430
- Callen, M., S. Gulzar, A. Rezaee, and J.N. Shapiro (2020). 'Choosing Ungoverned Space: The Removal of Pakistan's Frontier Crimes Regulation'.
- Cantoni, D., F. Hagemeister, and M. Westcott, M. (2019). Persistence and Activation of Right-Wing Political Ideology'. Rationality and Competition Discussion Paper 143. Available at: https://rationality-and-competition.de/wp-content/uploads/discussion\_paper/143.pdf (accessed 6 March 2023)
- Caroe, O. (1958). The Pathans, 550 BC-AD 1957. London: Macmillan & Co.
- Cattaneo, M.D., N. Idrobo, and R. Titiunik (2019). A Practical Introduction to Regression Discontinuity Designs: Foundations. Cambridge: Cambridge University Press. https://doi.org/10.1017/9781108684606
- Cederman, L. E., and M. Vogt (2017). Dynamics and Logics of Civil War'. *Journal of Conflict Resolution*, 61(9): 1992–2016. https://doi.org/10.1177/0022002717721385
- Chambru, C. (2019). 'Do the Right Thing! Leaders, Weather Shocks and Social Conflicts in Pre-Industrial France'. EHES Working Paper 161. Vienna: European Historical Economics Society
- Chauvin, E.D. (2016). 'State History and Contemporary Conflict: Evidence from Sub-Saharan Africa'. Instituto de Economía PUC Documentos de Trabajo 475. Santiago: IE-PUC.
- Chronopoulos, D.K., S. Kampanelis, D. Oto-Peralías, D., and J.O. Wilson (2021). 'Ancient Colonialism and the Economic Geography of the Mediterranean'. *Journal of Economic Geography*, 21(5): 717–59. https://doi.org/10.1093/jeg/lbaa028
- Ciccone, A. (2008). 'Transitory Economic Shocks and Civil Conflict'. CEPR Discussion Paper DP7081. Available at: https://ssrn.com/abstract=1344640 (accessed 10 February 2023).
- Ciolek, T. M. (1999). Old World Trade Routes (OWTRAD) Project. Canberra: Asia Pacific Research Online. Available at: http://www.ciolek.com/owtrad.html.
- Cleaver, F. (2002). Reinventing Institutions: Bricolage and the Social Embeddedness of Natural Resource Management'. *The European Journal of Development Research*, 14(2): 11–30. https://doi.org/10.1080/714000425
- Coll, S. (2004). Ghost Wars: The Secret History of the CLA, Afghanistan, and bin Laden, from the Soviet Invasion to September 10, 2001. London: Penguin.
- Collier, P., and A. Hoeffler (1998). 'On Economic Causes of Civil War'. Oxford Economic Papers, 50(4): 563–73. https://doi.org/10.1093/oep/50.4.563
- Collier, P., and A. Hoeffler (2004). 'Greed and Grievance in Civil War'. Oxford Economic Papers, 56(4): 563–95. https://doi.org/10.1093/oep/gpf064
- Collier, P., and A. Hoeffler (2007). 'Civil War'. In T. Sandler and K. Hartley (eds), *Handbook of Defense Economics, Volume 2, Defense in a Globalized World*. Amsterdam and Oxford: Elsevier, North Holland.
- Colona, F., and R. Jaffe (2016). 'Hybrid Governance Arrangements'. *The European Journal of Development* Research, 28(2): 175–83. https://doi.org/10.1057/ejdr.2016.5
- Constitution of Pakistan (1973). Part XII: Miscellaneous, Chapter 3: Tribal Areas, Articles 246-247'. Available at: https://www.pakistani.org/pakistan/constitution/part12.ch3.html (accessed 7 March 2023).
- Cuvelier, J., K. Vlassenroot, and N. Olin (2014). 'Resources, Conflict and Governance: a Critical Review'. *The Extractive Industries and Society*, 1(2): 340–50. https://doi.org/10.1016/j.exis.2014.07.006

- Dalgaard, C. J., N. Kaarsen, O. Olsson, and P. Selaya (2022). 'Roman Roads to Prosperity: Persistence and Non-Persistence of Public Infrastructure'. *Journal of Comparative Economics*, 50(4): 896–916. https://doi.org/10.1016/j.jce.2022.05.003
- Davis, H.W.C. (1926). 'The Great Game in Asia (1800-1844)'. Proceedings of the British Academy, 12: 227-256.
- Dell, M. (2010). 'The Persistent Effects of Peru's Mining Mita'. Econometrica, 78(6): 1863–903. https://doi.org/10.3982/ECTA8121
- Dell, M., and P. Querubin (2018). 'Nation Building through Foreign Intervention: Evidence from Discontinuities in Military Strategies'. *Quarterly Journal of Economics*, 133(2): 701–64. https://doi.org/10.1093/qje/qjx037
- Dell, M., N. Lane, and P. Querubin (2018). 'The Historical State, Local Collective Action, and Economic Development in Vietnam'. *Econometrica*, 86(6): 2083–121. https://doi.org/10.3982/ECTA15122
- Dell, M., and B. Olken. (2020). 'The Developmental Effects of the Extractive Colonial Economy: The Dutch Cultivation System in Java'. Review of Economic Studies, 87(1): 164–203. https://doi.org/10.1093/restud/rdz017
- Depetris-Chauvin, Emilio, State History and Contemporary Conflict: Evidence from Sub-Saharan Africa (October 25, 2015). Available at SSRN: https://doi.org/10.2139/ssrn.2679594.
- Depetris-Chauvin, E. and Ö. Özak. (2020). Borderline Disorder: (De facto) Historical Ethnic Borders and Contemporary Conflict in Africa'. Available at SSR. https://doi.org/10.2139/ssrn.3541025.
- Dincecco, M., Fenske, J., Menon, A. and Mukherjee, S. (2022). 'Pre-Colonial Warfare and Long-Run Development in India'. *The Economic Journal*, 132(643): 981–1010. https://doi.org/10.1093/ej/ueab089
- Donaldson, D. (2018). 'Railroads of the Raj: Estimating the Impact of Transportation Infrastructure'. American Economic Review, 4-5(108): 899–934. https://doi.org/10.1257/aer.20101199
- Dorril, S. (2002). MI6: Inside the Covert World of Her Majesty's Secret Intelligence Service. London: Simon and Schuster.
- Elahi, N. (2019). Terrorism in Pakistan: The Tehreek-e-Taliban Pakistan (TTP) and the Challenge to Security. London: Bloomsbury Publishing. https://doi.org/10.5040/9781350988392
- Embree, A.T. (ed.). (1977). Pakistan's Western Borderlands: The Transformation of a Political Order. Durham, NC: Carolina Academic Press.
- Engerman, S.L., and K.L. Sokoloff (1994). 'Factor Endowments: Institutions, and Differential Paths of Growth Among New World Economies: a View from Economic Historians of the United States'. Historical Paper 66. Cambridge, MA: National Bureau of Economic Research. https://doi.org/10.3386/h0066
- Fair, C., and S.G. Jones (2009). Pakistan's War Within'. *Survival*, 51(6): 161–88. https://doi.org/10.1080/00396330903465204
- Fair, C. C., N. Malhotra, and J.N. Shapiro (2010). Islam, Militancy, and Politics in Pakistan: Insights from a National Sample'. *Terrorism and Political Violence*, 22(4): 495–521. https://doi.org/10.1080/09546553.2010.492305
- Fearon, J.D. (2007). 'Economic Development, Insurgency, and Civil War'. In E. Helpman (ed.), Institutions and Economic Performance. Cambridge, MA: Harvard University Press. https://doi.org/10.2307/j.ctv21hrgnz.11
- Fearon, J.D., and D.D. Laitin (2003). 'Ethnicity, Insurgency, and Civil War'. American Political Science Review, 97(1): 75–90. https://doi.org/10.1017/S0003055403000534
- Fearon, J., and D. Laitin (2014). 'Does Contemporary Armed Conflict have Deep Historical Roots?'. SSRN 1922249. https://doi.org/10.2139/ssrn.1922249
- Fouka, V., and H.J. Voth (2013). 'Reprisals Remembered: German-Greek Conflict and Car Sales During the Euro Crisis'. Unpublished Manuscript.

- Frankema, E., J. Williamson, and P. Woltjer (2018). 'An Economic Rationale for the West African Scramble? The Commercial Transition and the Commodity Price Boom of 1835–1885'. *The Journal of Economic History*, 78(1): 231–67. https://doi.org/10.1017/S0022050718000128
- GAEZ Data Portal (2012). 'Global Agro-Ecological Zones'. Available at: https://gaez.fao.org/ (accessed 15 March 2022).
- Garrett, N., S. Sergiou, and K. Vlassenroot (2009). 'Negotiated Peace for Extortion: The Case of Walikale Territory in Eastern DR Congo'. *Journal of Eastern African Studies*, 3(1): 1–21. https://doi.org/10.1080/17531050802682671
- Gelman, A., and G. Imbens (2019). 'Why High-order Polynomials Should Not Be Used in Regression Discontinuity Designs'. *Journal of Business & Economic Statistics*, 37(3): 447–56. https://doi.org/10.1080/07350015.2017.1366909
- Gleditsch, K.S. (2007). "Transnational Dimensions of Civil War'. Journal of Peace Research, 44(3): 293–309. https://doi.org/10.1177/0022343307076637
- GTD (2021). 'Global Terrorism Database'. Available at: https://www.start.umd.edu/gtd/ (accessed 15 January 2021).
- Hagmann, T., and D. Péclard (2010). Negotiating Statehood: Dynamics of Power and Domination in Africa'. *Development and Change*, 41(4): 539–62. https://doi.org/10.1111/j.1467-7660.2010.01656.x
- Heldring, L. (2021). "The Origins of Violence in Rwanda'. *The Review of Economic Studies*, 88(2): 730–63. https://doi.org/10.1093/restud/rdaa028
- Hegre, H., and N. Sambanis (2006). 'Sensitivity Analysis of Empirical Results on Civil War Onset'. Journal of Conflict Resolution, 50(4): 508–35. https://doi.org/10.1177/0022002706289303
- Herbst, J. (2000). States and Power in Africa: Comparative Lessons in Authority and Control. Princeton, NJ: Princeton University Press.
- Hijmans, R.J., S.E. Cameron, J.L. Parra, P.G. Jones, and A. Jarvis (2005). 'Very High Resolution Interpolated Climate Surfaces for Global Land Areas'. *International Journal of Climatology: A Journal of* the Royal Meteorological Society, 25(15): 1965–78. https://doi.org/10.1002/joc.1276
- Hopkins, B.D. (2015). 'The Frontier Crimes Regulation and Frontier Governmentality'. The Journal of Asian Studies, 74(2): 369–89. https://doi.org/10.1017/S0021911815000030
- Hopkins, B.D. (2020). Ruling the Savage Periphery: Frontier Governance and the Making of the Modern State. Cambridge, MA: Harvard University Press. https://doi.org/10.4159/9780674246164
- Hopkins, B.D., and M. Marsden (2012). Fragments of the Afghan Frontier. New York, NY: Columbia University Press.
- Hopkirk, P. (2001). The Great Game: on Secret Service in High Asia. Oxford: Oxford University Press.
- Hudson, L., C. S. Owens, and M. Flannes (2011). 'Drone Warfare: Blowback from the New American Way of War'. *Middle East Policy*, 18(3): 122–32. https://doi.org/10.1111/j.1475-4967.2011.00502.x
- Hussain, R. (2005). Pakistan and the Emergence of Islamic Militancy in Afghanistan. Aldershot: Gower Publishing Ltd.
- HYDE (2006). 'History Database of the Global Environment'. Available at: https://www.pbl.nl/en/image/links/hyde (accessed 20 March 2022).
- Ibbetson, D. C. (1881). Census of the Panjab (Volume I). Calcutta: Government of India.
- Imperial Gazetteer of India (2005). *Provincial Series: North West Frontier Province*. Unabridged facsimile of edition published in 1908. Elibron Classics. Available at: https://www.amazon.com/Imperial-Gazetteer-India-Frontier-Provincial/dp/1402167989 (accessed 6 March 2023).
- Iyigun, M., N. Nunn, and N. Qian (2017). 'Winter is Coming: The Long-run Effects of Climate Change on Conflict, 1400-1900'. NBER Working Paper 23033. Cambridge, MA: National Bureau of Economic Research. https://doi.org/10.3386/w23033

- Jaques, T. (2007). Dictionary of Battles and Sieges: A Guide to 8,500 Battles from Antiquity through the Twenty-First Century. Westport, CT: Greenwood Press.
- Johnston, P.B., and A.K. Sarbahi (2016). 'The Impact of US Drone Strikes on Terrorism in Pakistan'. International Studies Quarterly, 60(2): 203–19. https://doi.org/10.1093/isq/sqv004
- Jones, S.G., and C.D. Fair (2010). *Counterinsurgency in Pakistan*. Arlington VA: National Security Research Division, RAND Corporation.
- Khan, K. (2013). 'Distributive Consideration in Institutional Change: The Case of Zia's Islamization Policy in Pakistan'. *Constitutional Political Economy*, 24(2): 139–65. https://doi.org/10.1007/s10602-013-9136-9
- Kocher, M. A., T. B. Pepinsky, and S. N. Kalyvas (2011). 'Aerial Bombing and Counterinsurgency in the Vietnam War.' American Journal of Political Science, 55(2): 201–18. https://doi.org/10.1111/j.1540-5907.2010.00498.x
- Kolsky, E. (2015). 'The Colonial Rule of Law and the Legal Regime of Exception: Frontier "Fanaticism" and State Violence in British India'. *The American Historical Review*, 120(4): 1218–46. https://doi.org/10.1093/ahr/120.4.1218
- Korting, C., C. Lieberman, J. Matsudaira, Z. Pei, and Y. Shen (2023). 'Visual Inference and Graphical Representation in Regression Discontinuity Designs'. *Quarterly Journal of Economics*, 138(3): 1977– 2019. https://doi.org/10.1093/qje/qjad011
- La Ferrara, E., and R.H. Bates (2001). 'Political Competition in Weak States'. *Economics & Politics*, 13(2): 159–84. https://doi.org/10.1111/1468-0343.00088
- Lee, A., and K. A. Schultz (2012), 'Comparing British and French Colonial Legacies: A Discontinuity Analysis of Cameroon', *Quarterly Journal of Political Science*, 7(4): 365–410. https://doi.org/10.1561/100.00011022
- Lowes, S., and E. Montero (2021). 'The Legacy of Colonial Medicine in Central Africa'. American Economic Review, 111(4): 1284–314. https://doi.org/10.1257/aer.20180284
- Lund, C. (2006). 'Twilight Institutions: Public Authority and Local Politics in Africa'. Development and Change, 37(4): 685-705. https://doi.org/10.1111/j.1467-7660.2006.00497.x
- Lynch III, T.F. (2018). "The Decades-Long "Double-Double Game". *Military Review*. Available at: https://www.armyupress.army.mil/Journals/Military-Review/English-Edition-Archives/July-August-2018/Lynch-Double/ (accessed 10 February 2023).
- Mahmood, R., and M. Jetter (2019). 'Military Intervention Via Drone Strikes'. IZA Discussion Paper 12318. Bonn: Institute of Labor Economics. https://doi.org/10.2139/ssrn.3390307
- Mahsud, N.H.K., M.S. Ahmad, and Wasai. (2021). 'Frontier Crimes Regulation (FCR): From Introduction to Abolition'. *Global Political Review*, VI: 109–20. https://doi.org/10.31703/gpr.2021(VI-I).10
- Mamdani, M. (1999). 'Historicizing Power and Responses to Power: Indirect Rule and Its Reform'. *Social Research*, 66(3): 859–86.
- Mamdani, M. (2018). 'Citizen and Subject: Contemporary Africa and the Legacy of Late Colonialism'. Princeton: Princeton University Press.
- McAllister, E., and L. Marsh (2021). 'Extremists Target African Village Leaders in Wave of Assassinations'. *Reuters*, 8 October. Available at: https://www.reuters.com/investigates/special-report/africa-islamists-sahel-leaders/ (accessed 10 February 2023).
- Meagher, K. (2012). "The Strength of Weak States? Non-State Security Forces and Hybrid Governance in Africa'. Development and Change, 43(5): 1073-101. https://doi.org/10.1111/j.1467-7660.2012.01794.x
- Meagher, K., T. De Herdt, and K. Titeca (2014). 'Unravelling Public Authority: Paths of Hybrid Governance in Africa'. Research Brief 10. Gent: IS Academy.
- Menkhaus, K. (2006). 'Governance Without Government in Somalia: Spoilers, State Building, and the Politics of Coping'. *International Security*, 31(3): 74–106. https://doi.org/10.1162/isec.2007.31.3.74

- Michalopoulos, S., and E. Papaioannou (2016). 'The Long-run Effects of the Scramble for Africa'. *American Economic Review*, 106(7): 1802–48. https://doi.org/10.1257/aer.20131311
- Michalopoulos, S., A. Naghavi, and G. Prarolo (2018). 'Trade and Geography in the Spread of Islam'. *The Economic Journal*, 128(616), 3210–41. https://doi.org/10.1111/ecoj.12557
- Miguel, E., S. Satyanath, and E. Sergenti (2004). 'Economic Shocks and Civil Conflict: An Instrumental Variables Approach'. *Journal of Political Economy*, 112(4): 725–53. https://doi.org/10.1086/421174
- Moscona, J., N. Nunn, and J.A. Robinson (2020). 'Segmentary Lineage Organization and Conflict in Sub-Saharan Africa'. *Econometrica*, 88(5): 1999–2036. https://doi.org/10.3982/ECTA16327
- Naseemullah, A. (2014). 'Shades of Sovereignty: Explaining Political Order and Disorder in Pakistan's Northwest'. *Studies in Comparative International Development*, 49(4): 501–22. https://doi.org/10.1007/s12116-014-9157-z
- Naseemullah, A. (2022). Patchwork States: The Historical Roots of Subnational Conflict and Competition in South Asia. Cambridge: Cambridge University Press. https://doi.org/10.1017/9781009158404
- Naseemullah, A., and P. Staniland (2016). Indirect Rule and Varieties of Governance'. *Governance*, 29(1): 13–30. https://doi.org/10.1111/gove.12129
- Nawaz. S. (2011). 'Learning by Doing: The Pakistan Army's Experience with Counterinsurgency'. Atlantic Council. Available at: https://www.atlanticcouncil.org/in-depth-research-reports/report/learning-by-doing-the-pakistan-army-s-experience-with-counterinsurgency/.
- New America (2021). 'The Drone War in Pakistan'. Available at: https://www.newamerica.org/international-security/reports/americas-counterterrorism-wars/thedrone-war-in-pakistan/ (accessed 7 August 2021).
- Nichols, R. (2001). The Frontier Crimes Regulation: A History in Documents. Oxford: Oxford University Press.
- Nielsen, R.A., M.G. Findley, Z.S. Davis, T. Candland, and D.L. Nielson (2011). 'Foreign Aid Shocks As a Cause of Violent Armed Conflict'. *American Journal of Political Science*, 55(2): 219–32. https://doi.org/10.1111/j.1540-5907.2010.00492.x
- Noor, S. (2006). 'Afghan Refugees After 9/11'. Pakistan Horizon, 59(1): 59-78.
- Nunn, N. (2009). "The Importance of History for Economic Development". *Annual Review of Economics*, 1(1): 65–92. https://doi.org/10.1146/annurev.economics.050708.143336
- Nunn, N. (2021). 'History as Evolution'. In A. Bisin and G. Federico (eds), *The Handbook of Historical Economics*. New York, NY: North Holland. https://doi.org/10.1016/B978-0-12-815874-6.00010-1
- Nunn, N., and D. Puga (2012). Ruggedness: The Blessing of Bad Geography in Africa'. Review of Economics and Statistics, 94(1): 20–36. https://doi.org/10.1162/REST\_a\_00161
- Ochsner, C., and F. Roesel (2017). 'Activated History-The Case of the Turkish Sieges of Vienna'. CESifo Working Paper 6586. http://dx.doi.org/10.2139/ssrn.3023816
- Oto-Peralías, D., and D. Romero-Ávila (2017). 'Historical Frontiers and the Rise of Inequality: The Case of the Frontier of Granada'. *Journal of the European Economic Association*, 15(1): 54–98. https://doi.org/10.1093/jeea/jvw004
- Paik, C., and K. Shahi (2023). 'Ancient Nomadic Corridors and Long-run Development in the Highlands of Asia.' *Explorations in Economic History*, 89: 101482. https://doi.org/10.1016/j.eeh.2022.101482
- Pant, S. (2018). "The Frontier Crimes Regulation in Colonial India: Local Critiques and Persistent Effects'. South Asia: Journal of South Asian Studies, 41(4): 789–805. https://doi.org/10.1080/00856401.2018.1531470
- Pentz, P.A. (1987). 'The Mujahidin Middleman: Pakistan's Role in the Afghan Crisis and the International Rule of Non-Intervention'. *Penn State Journal of International Law*, 6(3): 377.
- Pierson, P. (2004). Politics in Time: History, Institutions, and Social Analysis. Princeton: Princeton University Press. https://doi.org/10.1515/9781400841080

- Popescu, B.G. (2023). "The Developmental Legacies of Border Buffer Zones: The Case of Military Colonialism." *Journal of Historical Political Economy*, 3(1): 31–63. https://doi.org/10.1561/115.00000045
- Raeymaekers, T., K. Menkhaus, and K. Vlassenroot (2008). 'State and Non-state Regulation in African Protracted Crises: Governance Without Government?'. *Afrika Focus*, 21(2): 7–21. https://doi.org/10.1163/2031356X-02102003
- Rana, M.A. (2004). A to Z of Jehadi Organizations in Pakistan. Lahore: Mashal Publishing, Ltd.
- Rehman, F. U., M. Nasir, and M. Shahbaz. (2017). 'What Have We Learned? Assessing the Effectiveness of Counterterrorism Strategies in Pakistan', *Economic Modelling*, 64(August): 487–95. https://doi.org/10.1016/j.econmod.2017.02.028
- Reid, R.J. (2012). Warfare in African History (Vol. 6). Cambridge: Cambridge University Press. https://doi.org/10.1017/CBO9781139043090
- Reuters (2006). Pakistani Leader Claims U.S. Threat after 9/11'. *The New York Times*, 22 September. Available at: https://www.nytimes.com/2006/09/22/world/asia/22pakistan.html (accessed 13 February 2023).
- Riedel, B. (2014). What We Won: America's Secret War in Afghanistan, 1979-89. Washington, DC: Brookings Institution Press.
- Riley, S.J., S.D. DeGloria, and R. Elliot (1999). 'Index that Quantifies Topographic Heterogeneity'. *Intermountain Journal of Sciences*, 5(1–4): 23–27.
- Rodrik, D. (1999). 'Where Did All the Growth Go? External Shocks, Social Conflict, and Growth Collapses'. *Journal of Economic Growth*, 4(4): 385–412. https://doi.org/10.1023/A:1009863208706
- Roitman, J. (2005). Fiscal Disobedience. An Anthropology of Economic Regulation in Central Africa. Princeton, NJ: Princeton University Press. https://doi.org/10.1515/9780691187044
- Salehyan, I., and K.S. Gleditsch (2006). 'Refugees and the Spread of Civil War'. *International Organization*, 60(2): 335–66. https://doi.org/10.1017/S0020818306060103
- Sambanis, N. (2005). 'Using Case Studies to Refine and Expand the Theory of Civil War'. In P. Collier and N. Sambanis (eds), *Understanding Civil War: Evidence and Analysis*. Washington, DC: World Bank.
- SATP (2021). 'Datasheet–Pakistan'. South Asia Terrorism Portal. Available at: https://www.satp.org/ (accessed 10 January 2021).
- Savun, B., and D.C. Tirone (2011). 'Foreign Aid, Democratization, and Civil Conflict: How Does Democracy Aid Affect Civil Conflict?'. *American Journal of Political Science*, 55(2): 233–46. https://doi.org/10.1111/j.1540-5907.2010.00501.x
- Scott, J, C. (2009). The Art of Not Being Governed. New Haven: Yale University Press.
- Shinwari, N.A. (2011). Understanding Jirga: Legality and Legitimacy in Pakistan's Federally Administered Tribal Areas. Islamabad: Community Appraisal & Motivation Programme (CAMP).
- Shinwari, N.A. (2012). Understanding FATA: Attitudes Towards Governance, Religion & Society in Pakistan's Federally Administered Tribal Areas. Volume V. Islamabad: Community Appraisal & Motivation Programme (CAMP).
- Shuttle Radar Topography Mission (2018). 'SRTM 90m DEM Digital Elevation Database'. Available at: https://srtm.csi.cgiar.org/ (accessed 11 March 2022).
- Siddiqui, Z. (2018). 'Frontier Crimes Regulation: A Past that Never Ends'. *Dawn*, 15 May. Available at: https://www.dawn.com/news/1407020 (accessed 13 February 2023).
- Skaperdas, S. (2008). 'An Economic Approach to Analyzing Civil Wars'. Economics of Governance, 9(1): 25– 44. https://doi.org/10.1007/s10101-007-0043-2
- Stewart, J. (2007). The Savage Border: The Story of the North-West Frontier. Stroud: Sutton Publishing.

- Tanguay-Renaud, F. (2002). 'Post-Colonial Pluralism, Human Rights & (and) the Administration of Criminal Justice in the Federally Administered Tribal Areas of Pakistan'. Singapore Journal of International and Company Law, 6: 541–596.
- Tripodi, C. (2009). 'British Policy on the North-West Frontier of India 1877-1947: A Suitable Precedent for the Modern Day?'. Commentary. Royal United Services Institute.
- Tripodi, C. (2020). The Unknown Enemy: Counterinsurgency and the Illusion of Control. Cambridge: Cambridge University Press. https://doi.org/10.1017/9781108341110
- Turk, M.A. (2021). 'The Dilemma of Justice System in Formerly FATA of Pakistan: Challenges and Prospects'. *Journal of Law and Social Policy*, 3. Available at: https://www.bahria.edu.pk/buic/law/wpcontent/uploads/2021/10/The-Dilemma-of-Justice-System-in-Formerly-FATA-of-Pakistan-Challenges-and-Prospects.pdf (accessed 6 March 2023).
- Ullah, A., and S.U. Hayat (2017). 'The Recent Electoral Reforms in Federally Administered Tribal Areas (FATA): An Appraisal'. *Journal of Political Studies*, 24(2).
- UCD (2023) 'About UCDP'. Uppsala Conflict Data Program. Available at: https://www.pcr.uu.se/research/ucdp/about-ucdp/ (accessed 31 January 2023).
- Usmani, Z. (2017). 'Pakistan Body Count', Drone Attacks Dataset, Kaggle Dataset Repository, 25 January 2017. Available at: https://www.kaggle.com/datasets/zusmani/pakistandroneattacks?resource=download.
- Voors, M.J., and E.H. Bulte (2014). 'Conflict and the Evolution of Institutions: Unbundling Institutions at the Local Level in Burundi'. *Journal of Peace Research*, 51(4): 455–69. https://doi.org/10.1177/0022343314531264
- Watts, S., J. H. Campbell, P. B. Johnson, S. Lalwani, and S. H. Bana (2014). 'Counterinsurgency in Pakistan'. In J. H. Campbell, P. B. Johnson, S. Lalwani, and S. H. Bana, *Countering Others' Insurgencies: Understanding U.S. Small-Footprint Interventions in Local Context.* Rand Corporation.
- Weinbaum, M.G. (1991). 'War and Peace in Afghanistan: The Pakistani Role'. *Middle East Journal*, 45(1): 71–85.
- White, J.T. (2008). 'The Shape of Frontier Rule: Governance and Transition, from the Raj to the Modern Pakistani Frontier'. *Asian Security*, 4(3): 219–43. https://doi.org/10.1080/14799850802306328
- Wig, T. (2016). 'Peace from the Past: Pre-Colonial Political Institutions and Civil Wars in Africa'. *Journal of Peace Research*, 53(4): 509–24. https://doi.org/10.1177/0022343316640595
- Wimmer, A., and B. Min (2006). 'From Empire to Nation-state: Explaining Wars in the Modern World, 1816–2001'. American Sociological Review, 71(6): 867–97. https://doi.org/10.1177/000312240607100601
- WorldClim (2020). 'Global Climate and Weather Data'. Available at: https://worldclim.org/ (accessed 13 March 2022).
- Zahab, M.A. (2013). 'Kashars Against Mashars: Jihad and Social Change in the FATA'. In B.D. Hopkins and M. Marsden (eds), *Beyond Swat*. New York, NY: Columbia University Press.

# Tables

Table 1: Historical institutional discontinuity-key illustrations

Frontier agencies	Settled districts	
Political agent as the sole official representative of the colonial administration. These were often ex-military officials who were responsible for managing the relationship between the colonial state and tribes	Each district with a Deputy Commissioner supported by <i>Tahsildar</i> s and <i>Naib Tahsildars</i> invested with criminal, civil and revenue powers	Bureau
	<i>Field Kanungos</i> (revenue officials) responsible for supervising <i>Patwaris</i> (accountants)	cratic admir
	<i>Patwaris</i> (accountants) responsible for maintaining revenue records of between 4 to 5 villages	nistration
	Chaukidars (village headmen) responsible for collective revenue	J
Jirga system (council of tribal elders) used for dispute resolution	Divisional Courts established under regulation VII of 1901	Justic
Criminal justice was governed through the Frontier Crimes Regulation (FCR), introduced in 1872.	District Courts established under regulation VII of 1901	e system

	Rural District Boards constituted under Act XX of 1883		Represe
No electoral representation allowed until 1997 when only local elites (Maliks) were allowed to vote.	Municipal Boards constituted under Act XV of 1867		entative institu
	Provincial legislature constituted under the Government of India Act 1935. Elections to the legislature held in 1937 on a restricted franchise basis with separate electorates for Muslims and Non- Muslims. The main political parties contesting was the Congress, the Muslim Nationalist party, the Hindu–Sikh Nationalist Party (HSNP), and the Muslim Independent Party (MIP).		tions
From 1890s onwards <i>Khassadar</i> s working under <i>Maliks</i> for border protection	Administration of the civil police force in the settled districts vested in an Inspector-General	]	
Power to 'arrest', 'call for tribal jirgahs' (councils) and 'dispense justice on the spot'	Each district under a Superintendent working under the general control of the Deputy Commissioner	-	Policing & secur
From 1915 onwards <i>Khassadars</i> supported by the Frontier Constabulary, responsible for patrolling the border between Frontier Agencies and Settled Districts	Police jurisdiction based on <i>thanas</i> (units of civil police administration), each under the control of a sub-inspector		ity

Source: authors' elaboration based on Imperial Gazetteer of India: Provincial series, North-west Frontier Province, 1908; Nichols (2001); Kolsky (2015).

#### Table 2: Frontier rule—analysis of mean differences

	Whole sample			Within 50km of FR border		
	FR	Non-FR	Mean difference	FR	Non-FR	Mean difference
	Mean	values		Mean		
	(1)	(2)	(3)	(4)	(5)	(6)
Conflict incidents against the state	1.460	0.098	1.362***	1.779	0.177	1.602***
	656	8567	(0.223)	484	634	(0.301)
In(1+Conflict incidents against the state)	0.367	0.033	0.335***	0.417	0.074	0.343***
	656	8567	(0.030)	484	634	(0.039)
Deaths in incidents against the state	4.212	0.178	4.034***	4.911	0.538	4.373***
	656	8567	(0.595)	484	634	(0.791)
In(1+Deaths in incidents against the state)	0.511	0.031	0.480***	0.568	0.091	0.476***
	656	8567	(0.044)	484	634	(0.056)
Injuries in incidents against the state	3.848	0.261	3.586***	4.789	0.661	4.128*
	656	8567	(0.601)	484	634	(0.834)
In(1+Injuries in incidents against the state)	0.482	0.034	0.447***	0.555	0.095	0.460***
	656	8567	(0.042)	484	634	(0.056)

Note: the unit of observation is a 10km-by-10km grid cell. Columns 1–3 are based on the full sample of grid cells that comprise Pakistan, whereas columns 4–6 restrict the sample to a 50 km buffer zone either side of the FR border. Columns 1–2 report the mean of each conflict measure between the FR and non-FR grid cells for the full sample. Columns 4 and 5 report the mean of each conflict measure between the FR and non-FR grid cells for the FR border. Finally, columns 3 and 6 show the result for a two-sample t-test for difference in means in each of the conflict measure between the FR and non-FR grid cells. The number of observations is in italics. The standard errors are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Table 3: Balance on geographic, climatic, and historic characteristics

	Sample: observations within 50 km from FR border									
	Linear running variable in Euclidean distance to the border									
Dependent variable:	Ruggedness	Slope	Topography	Precipitation (mean)	Precipitation (std dev)	Temperature	Pre-FR conflict	Pre-FR pop density	Wheat suitability	
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	
Inside FR border	-19.150	-0.028	2.375	-7.722	1.154	0.059	-0.001	0.023	83.215	
	(16.183)	(0.255)	(1.875)	(5.133)	(1.715)	(0.245)	(0.013)	(0.060)	(99.891)	
Observations	1,106	1,106	1,106	1,118	1,118	1,118	1,118	1,118	1,109	
95% C.I.	17.125]	[648 ; .520]	6.765]	[-22.022 , 1.872]	[-2.842 , 4.986]	[405 ; .696]	[030 ; .023]	[125 ; .135]	336.41]	
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
BW-type	Cerrd	Cerrd	Cerrd	Cerrd	Cerrd	Cerrd	Cerrd	Cerrd	Cerrd	
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular	
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	

Note: the unit of observation is a 10km-by-10km grid cell. Columns 1–8 report the RD estimates for geographic, climatic, agricultural, and historic variables within a 50 km buffer zone of the FR border. All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. The standard errors are reported in parentheses and are clustered at the 20 km border segment level. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

#### Table 4: Frontier rule and conflict against the state

	Sample: observations within 50 km from FR border										
		Linear running variable in Euclidean distance to the border									
Dependent variable:	In(1+incidents	s against state)	In(1+deaths in inci	dents against state)	In(1+injuries in incidents against state)						
	(1)	(2)	(3)	(4)	(5)	(6)					
Inside FR border	0.297***	0.318***	0.312***	0.210***	0.708***	0.684***					
	(0.054)	(0.054)	(0.076)	(0.081)	(0.078)	(0.078)					
Observations	1,118	1,105	1,118	1,105	1,118	1,105					
95% C.I.	[.228 ; .471]	[.249 ; .501]	[.194 ; .536]	[.078 ; .437]	[.599 ; .963]	[.579 ; .947]					
Controls	No	Yes	No	Yes	No	Yes					
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes					
BW-type	mserd	mserd	mserd	mserd	mserd	mserd					
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular					
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID					

Note: the unit of observation is a 10km-by-10km grid cell. In columns 1 and 2, the outcome variable is the number of conflict incidents against the state; in columns 3 and 4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5 and 6, the dependent variable is the number of injuries in conflict incidents against the state; and in columns 5 and 6, the dependent variable is the number of injuries in conflict incidents against the state; and in columns 5 and 6, the dependent variable is the number of injuries in conflict incidents against the state; all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

	Linear running variable in Euclidean distance to the border									
Dependent variable:	In(1+incidents	s against state)	In(1+deaths in inci	dents against state)	In(1+injuries in inci	In(1+injuries in incidents against state)				
	(1)	(2)	(3)	(4)	(5)	(6)				
		I	Panel A: observations wit	thin 60 km from FR borde	er					
Inside FR border	0.281***	0.298***	0.299***	0.189**	0.612***	0.565***				
	(0.053)	(0.053)	(0.077)	(0.084)	(0.071)	(0.071)				
Observations	1,288	1,271	1,288	1,271	1,288	1,271				
95% C.I.	[.207 ; .438]	[.230 ; .466]	[.178 ; .514]	[.033 ; .409]	[.520 ; .833]	[.472 ; .788]				
		I	Panel B: observations wit	thin 40 km from FR borde	er					
Inside FR border	0.331***	0.347***	0.439***	0.315***	0.676***	0.715***				
	(0.056)	(0.056)	(0.080)	(0.075)	(0.077)	(0.079)				
Observations	933	924	933	924	933	924				
95% C.I.	[.231 ; .499]	[.250 ; .524]	[.313 ; .695]	[.197 ; .543]	[.561 ; .929]	[.591 ; .972]				
Controls	No	Yes	No	Yes	No	Yes				
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes				
BW-type	mserd	mserd	mserd	mserd	mserd	mserd				
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular				
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID				

Note: the unit of observation is a 10km-by-10km grid cell. In Panel A the regression sample is restricted to within 60 km of the FR border. Panel B restricts the sample to within 40 km of the FR border. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state; all parameterized as In(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

		:	Sample: observations wit	hin 50 km from FR bord	er					
	Linear running variable in Euclidean distance to the border									
Dependent variable:	In(1+incidents	s against state)	In(1+deaths in incid	dents against state)	In(1+injuries in inc	In(1+injuries in incidents against state)				
	(1)	(2)	(3)	(4)	(5)	(6)				
			Panel A: 18 km l	border segments						
Inside FR border	0.228***	0.307***	0.400***	0.416***	0.520***	0.578***				
Dependent variable: Inside FR border Observations 95% C.I. Inside FCR border Observations 95% C.I. Controls Segment FE BW-type	(0.060)	(0.056)	(0.087)	(0.086)	(0.079)	(0.084)				
Observations	1,118	1,105	1,118	1,105	1,118	1,105				
95% C.I.	[.136 ; .400]	[.222 ; .487]	[.266 ; .655]	[.286 ; .675]	[.400 ; .751]	[.440 ; .834]				
			Panel B: 15 km l	border segments						
Inside FCR border	0.114**	0.120**	0.139*	0.222***	0.434***	0.461***				
	(0.054)	(0.054)	(0.084)	(0.079)	(0.071)	(0.072)				
Observations	1,118	1,105	1,118	1,105	1,118	1,105				
95% C.I.	[.012 ; .268]	[.006 ; .274]	[012 ; .366]	[.074 ; .453]	[.312 ; .665]	[.344 ; .703]				
Controls	No	Yes	No	Yes	No	Yes				
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes				
BW-type	mserd	mserd	mserd	mserd	mserd	mserd				
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular				
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID				

Table 6: Frontier rule and conflict against the state using alternative border segments

Note: the unit of observation is a 10km-by-10km grid cell. In Panel A the regressions use 18 km border segment fixed effects. The regressions in Panel B use 15 km border segment fixed effects. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Table 7: Frontier rule and conflict against the state using Uppsala Conflict Data

		Sample: observations wit	hin 50 km from FR border						
	Linear running variable in Euclidean distance to the border								
Dependent variable:	In(1+confli	ct incidents)	In(1+deaths in c	conflict incidents)					
	(1)	(2)	(3)	(4)					
Inside FR border	0.219**	0.336***	0.376***	0.505***					
	(0.088)	(0.082)	(0.144)	(0.138)					
Observations	1,118	1,105	1,118	1,105					
95% C.I.	[.038 ; .463]	[.184 ; .587]	[.069 ; .774]	[.227 ; .913]					
Controls	No	Yes	No	Yes					
Segment FE	Yes	Yes	Yes	Yes					
BW-type	mserd	mserd	mserd	mserd					
Kernel	Triangular	Triangular	Triangular	Triangular					
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID					

Note: the unit of observation is a 10km-by-10km grid cell. In columns 1–2, the outcome variable is the number of conflict incidents; and in columns 3–4, the dependent variable is the number of deaths in conflict incidents, all parameterized as In(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2 and 4 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses.\*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Table 8: Frontier rule and conflict ac	ainst the state after excluding o	arid cells verv	close to the FR boundary

		:	Sample: observations wit	hin 50 km from FR bord	er						
		Linear running variable in Euclidean distance to the border									
Dependent variable:	In(1+incidents	s against state)	In(1+deaths in inci	dents against state)	In(1+injuries in incidents against state)						
	(1)	(2)	(3)	(4)	(5)	(6)					
Inside FR border	0.181***	0.165***	0.320***	0.239***	0.543***	0.436***					
	(0.060)	(0.058)	(0.093)	(0.087)	(0.089)	(0.086)					
Observations	1,099	1,086	1,099	1,086	1,099	1,086					
95% C.I.	[.053 ; .364]	[.034 ; .343]	[.162 ; .610]	[.162 ; .610] [.089 ; .514]		[.294 ; .739]					
Controls	No	Yes	No	Yes	No	Yes					
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes					
BW-type	mserd	mserd	mserd	mserd	mserd	mserd					
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular					
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID					

Note: the unit of observation is a 10km-by-10km grid cell. The regression sample for columns 1–6 drops grid cells that are very close (i.e. <= 0.5 km) to the FR border. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Table 9: Frontier rule and conflict against the state in the Pre- and Post-9/11 era

					Sample: obs	ervations wit	hin 50 km fro	om FR border					
	Linear running variable in Euclidean distance to the border												
			Pre	-9/11					Post	-9/11			
Depend ent variable:	In(1+incidents against i state)		In(1+de incident sta	In(1+deaths in In(1+in) incidents against incidents state) sta		juries in s against In(1+incident ate) state		In(1+deaths in dents against incidents against state) state)		eaths in s against ate)	In(1+injuries in incidents against state)		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)	
Inside FR													
border	0.000	-0.000	0.000	-0.001	0.000	-0.000	0.298***	0.318***	0.311***	0.210***	0.708***	0.684***	
	(0.000)	(0.007)	(0.000)	(0.009)	(0.000)	(0.006)	(0.054)	(0.054)	(0.076)	(0.081)	(0.078)	(0.078)	
Obs.	1,118	1,105	1,118	1,105	1,118	1,105	1,118	1,105	1,118	1,105	1,118	1,105	
95% C.I.	[002 ; .005]	[017 ; .022]	[002 ; .007]	[022 ; .027]	[001 ; .004]	[014 ; .018]	[.229 ; .472]	[.248 ; .501]	[.193 ; .535]	[.078 ; .437]	[.599 ; .964]	[.579 ; .947]	
Controls Segmen	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	
t FĔ	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	
BW-type	mserd Triangula	mserd Triangula	mserd Triangula	mserd Triangula	mserd Triangula	mserd Triangula	mserd Triangula	mserd Triangula	mserd Triangula	mserd Triangula	mserd Triangula	mserd Triangula	
Kernel	r	r	r	r	r	r	r	r	r	r	r	r	
ng	Segment _ID	Segment _ID	Segment _ID	Segment _ID	Segment _ID	Segment _ID	Segment	Segment _ID	Segment _ID	Segment _ID	Segment _ID	Segment _ID	

Note: the unit of observation is a 10km-by-10km grid cell. Columns 1–6 restrict the sample to the period prior to 9/11 from 1970 to 2000 and columns 7–12 restrict the sample to the period after 9/11 from 2001 to 2018. In columns 1–2 and 7–8, the outcome variable is the number of conflict incidents against the state; in columns 3–4 and 9–10, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6 and 11–12, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, 6, 8, 10, and 12 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

#### Table 10: Frontier rule and avenues of conflict management

	(1)	(2)	(3)	(4)	(5)	(6)
VARIABLES	Recours	e to MNA	No contact with MN	NA in last 2 months	Recours	e to jirga
FR status dummy	-0.016***	-0.019***	0.027***	0.011***	0.601***	0.542***
	(0.002)	(0.003)	(0.003)	(0.004)	(0.020)	(0.021)
Observations	6,030	6,030	6,030	6,030	6,030	6,030
95% C.I.	[0198;0132]	[0248;0126]	[.0210; .0334]	[.0035; .0191]	[.5612; .6411]	[.5002; .5838]
Controls	No	Yes	No	Yes	No	Yes
Adjusted R-squared	0.00108	0.00471	0.00175	0.02180	0.14700	0.20100

Note: the unit of observation is an individual. The explanatory variable is a dummy for whether an individual resides in a household that is inside the FR boundary. Dependent variables are a dummy for MNA being the main recourse for dispute resolution (columns 1–2), a dummy for contact with MNA in last two months (columns 3–4) and a dummy for Jirga being main recourse for dispute resolution (columns 5–6). The control variables are a dummy for gender, a dummy for educational status, ln(age), a dummy for hh source of income, a dummy for h monthly income range and a locality fixed effects. Robust standard errors are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Table 11: F	rontier rule	and trust in	state	institutions
-------------	--------------	--------------	-------	--------------

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
VARIABLES	Low trust in	parliament	Low trust in	district court	Low trust ir	n high court	Low trust in supreme court	
FR status dummy	0.116***	0.152***	0.123***	0.161***	0.195***	0.207***	0.196***	0.206***
	(0.021)	(0.022)	(0.021)	(0.023)	(0.022)	(0.024)	(0.022)	(0.024)
Observations	6,030	6,030	6,030	6,030	6,030	6,030	6,030	6,030
95% C.I.	[.0754; .1559]	[.1077; .1959]	[.0814; .1648]	[.1158; .2064]	[.1528; .2380]	[.1608; .2536]	[.1511; .2310]	[.1587; .2539]
Controls	No	Yes	No	Yes	No	Yes	No	Yes
Adjusted R-squared	0.00403	0.0249	0.00439	0.0262	0.0105	0.0312	0.0105	0.0307

Note: the unit of observation is an individual. The explanatory variable is a dummy for whether an individual resides in a household that is inside the FR boundary. Dependent variables are a dummy for very little to no trust in parliament (columns 1–2), a dummy for very little to no trust in the district court (columns 3–4), a dummy for very little to no trust in the high court (columns 5–6) and a dummy for very little to no trust in the supreme court (columns 7–8). The control variables are a dummy for gender, a dummy for educational status, ln(age), a dummy for hh source of income, a dummy for hh monthly income range and a locality fixed effects. Robust standard errors are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

#### Table 12: Frontier rule and targeting of tribal elders

	Sample: observations within 50 km from FR border										
		Linear running variable in Euclidean distance to the border									
Dependent variable:	In(1+incidents	against elders)	In(1+deaths in incid	lents against elders)	In(1+injuries in incid	lents against elders)					
	(1)	(2)	(3)	(4)	(5)	(6)					
Inside FR border	0.112***	0.098***	0.082**	0.071*	0.078**	0.072*					
	(0.033)	(0.032)	(0.038)	(0.037)	(0.039)	(0.039)					
Observations	1,118	1,105	1,118	1,105	1,118	1,105					
95% C.I.	[.059 ; .211]	[.043 ; .188]	[.018 ; .201]	[001 ; .178]	[.001 ; .197]	[008 ; .187]					
Controls	No	Yes	No	Yes	No	Yes					
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes					
BW-type	mserd	mserd	mserd	mserd	mserd	mserd					
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular					
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID					

Note: the unit of observation is a 10km-by-10km grid cell. In columns 1–2, the outcome variable is the number of conflict incidents targeting tribal elders; in columns 3–4, the dependent variable is the number of deaths in incidents targeting tribal elders; and in columns 5–6, the dependent variable is the number of injuries in incidents against the tribal elders, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat sutability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

			Sample: observat Linear running variab	ions within 50 km from FR borde le in Euclidean distance to the bo	r order	
Dependent variable:	In(1+incidents agai	nst state)	In(1+deaths in inci	idents against state)	In(1+injuries in inc state)	idents against
	(1)	(2)	(3)	(4)	(5)	(6)
Inside FR border	0.297***	0.316***	0.312***	0.219***	0.708***	0.686***
	(0.054)	(0.054)	(0.076)	(0.081)	(0.078)	(0.078)
Observations	1,118	1,105 [.246 ;	1,118	1,105	1,118	1,105
95% C.I.	[.228 ; .471]	.498]	[.194 ; .536]	[.086 ; .445]	[.599 ; .963]	[.579 ; .948]
Controls	No	Yes	No	Yes	No	Yes
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes
BW-type	mserd	mserd	mserd	mserd	mserd	mserd
Kernel	Triangular	Triangular Segment	Triangular	Triangular	Triangular	Triangular
Clustering	Segment_ID	_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID

Table 13: Frontier rule and conflict against the state including distance to Durand Line as an additional control

Note: the unit of observation is a 10km-by-10km grid cell. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: **distance to the** *Durand Line*, ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Table 14: Frontier rule and conflict against the state excluding grid cells contiguous to the Durand Line
---

		S	Sample: observations wit	hin 50 km from FR bord	er					
	Linear running variable in Euclidean distance to the border									
Dependent variable:	In(1+incidents	s against state)	In(1+deaths in inci	dents against state)	In(1+injuries in inci	In(1+injuries in incidents against state)				
	(1)	(2)	(3)	(4)	(5)	(6)				
Inside FR border	0.272***	0.279***	0.387***	0.299***	0.612***	0.562***				
	(0.053)	(0.053)	(0.075)	(0.075)	(0.071)	(0.070)				
Observations	1,070	1,065	1,070	1,065	1,070	1,065				
95% C.I.	[.200 ; .435]	[.210 ; .448]	[.277 ; .625]	[.176 ; .535]	[.521 ; .838]	[.469 ; .785]				
Controls	No	Yes	No	Yes	No	Yes				
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes				
BW-type	mserd	mserd	mserd	mserd	mserd	mserd				
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular				
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID				

Note: the unit of observation is a 10km-by-10km grid cell. The regression sample for columns 1–6 **drops grid cells that are contiguous to the Afghan Border (***Durand Line***). The sample, therefore, excludes areas where local spillovers of conflict from the Afghan side is most lilkely to have happened. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat sutability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.** 

Table 15: Conflict incidents against the state by specific origin

	197	0–2018	197	0–2000	2001–18	
Militant outfit	Count	Count Per cent		Count Per cent		Per cent
		Local			•	
Tehrik-i-Taliban Pakistan (TTP)	326	17.87	0	0.00	326	18.26
Baluchistan-based militants	216	11.84	0	0.00	216	12.10
Political militants wings	12	0.66	5	12.82	7	0.39
Lashkar-e-Jhangvi	9	0.49	0	0.00	9	0.50
Different local jihadi organization	79	4.33	5	12.82	74	4.15
Other militant organizations	27	1.48	2	5.13	25	1.40
		Foreign				
Haqqani network	0	0.00	0	0.00	0	0.00
Al-Qaida	11	0.60	0	0.00	11	0.62
		Unknown				
Unknown	1144	62.72	27	69.23	1117	62.58
Total	1824	100	39	100	1785	100

Note: the **Different local jihadi organizations** category includes groups like the Sipah-e-Sahaba, Hizb-I-Islami, Tehrik-e-Nafaz-e-Shariat-e-Mohammadi, Lashkar-e-Islam, Ansarul Islam, Jaish-e-Islam, Jaish al-Umar, Jamaat-ul-Ahrar, Harkatul Jihad-e-Islami, and so forth. Similarly, the **Other militant organizations** category consists of groups like Abdullah Azzam Brigades, Qari Kamran Group, Jundallah, Halqa-e-Mehsud, Hafiz Gul Bahadur Group, Khorasan, etc. **Unknown** includes those attacks that were not claimed by any terrorist organization.

Table	16: Frontie	r rule and	conflict a	against the	state	excluding	grid	cells w	here t	here ha	as been	a drone	strike

		Sample: observations within 50 km from FR border									
		Linear running variable in Euclidean distance to the border									
Dependent variable:	In(1+incidents	s against state)	In(1+deaths in inci	dents against state)	In(1+injuries in inci	In(1+injuries in incidents against state)					
	(1)	(2)	(3)	(4)	(5)	(6)					
Inside FR border	0.179***	0.172***	0.320***	0.271***	0.481***	0.393***					
	(0.053)	(0.053)	(0.072)	(0.073)	(0.064)	(0.063)					
Observations	1,059	1,046	1,059	1,046	1,059	1,046					
95% C.I.	[.096 ; .330]	[.089 ; .323]	[.202 ; .539]	[.144 ; .493]	[.387 ; .675]	[.294 ; .579]					
Controls	No	Yes	No	Yes	No	Yes					
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes					
BW-type	mserd	mserd	mserd	mserd	mserd	mserd					
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular					
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID					

Note: the unit of observation is a 10km-by-10km grid cell. The regression sample for columns 1-6 drops grid cells where there has ever been a drone strike. In columns 1-2, the outcome variable is the number of conflict incidents against the state; in columns 3-4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5-6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, pre-FR population density, and road density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Source: based on authors' calculations.

Table 17: Frontier rule and public infrastructure provision

	Sample: observations within 60 km from FR border										
	Linear running variable in Euclidean distance to the border										
Dependent variable:	health sites per 10000 persons	In(1+road length in km)	ln(1+waterway length in km)	In(1+colonial rail length in km)	In(1+distance to mughal roads in km)	In(1+distance to islamic trade and pilgrimage routes in km)					
	(1)	(2)	(3)	(4)	(5)	(6)					
Inside FR border	-0.103	-0.073	-0.200	0.009	-0.026	0.020					
	(0.168)	(0.141)	(0.130)	(0.038)	(0.017)	(0.022)					
Observations	1,288	1,288	1,288	1,288	1,288	1,288					
95% C.I.	[473 ; .251]	[328 ; .302]	[467 ; .110]	[077 ; .096]	[064 ; .010]	[029 ; .066]					
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes					
BW-type	cerrd	cerrd	cerrd	cerrd	cerrd	cerrd					
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular					
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID					

Note: the unit of observation is a 10km-by-10km grid cell. The outcome in column 1 is the number of health sites per 10,000 persons in a contemporary year (2017). In columns 2–3, the outcomes are the length of roads in km and the length of waterways in km, both measured in a pre-9/11 year (1992). Column 4 includes the length of colonial era railroads in km. Finally, in columns 5–6 we include historical (pre-FR rule) measures of underdevelopment: distance to major mughal era roads (1556–1707) and distance to islamic trade and pilgrimage routes (1300–1600). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

# Appendix A

# Conflict variables definition and data sources

# Terrorism

The Global Terrorism Database (GTD) (2021) defines 'a terrorist incident if it fulfils the following three criteria: (i) the incident must be intentional; (ii) the incident must entail some level of violence or threat of violence; and (iii) the perpetrators of the incidents must be sub-national actors. In addition, at least two of the following three criteria must be present for an incident to be included in the GTD: (i) the act must be aimed at attaining a political, economic, religious, or social goal; (ii) there must be evidence of an intention to coerce, intimidate or convey some other message to a larger audience (or audiences) than the immediate victims; and (iii) the action must be outside the context of legitimate warfare activities'.

# State attacks (the broader measure)

It consists of terrorist incidents targeting government officials and property, including attacks against civil servants, teachers, doctors, judges, police, military personnel, parliamentarians, educational institutions, healthcare facilities, courts, roads, bridges, airports, electricity grids, and gas installations.

# Non-state attacks (the broader measure)

It comprises attacks directed at both private property and individuals. This encompasses attacks on private individuals, including men, women, and children, as well as private residences, businessmen, businesses, religious figures and institutions, journalists, tourists, minorities, and more.

# Role of segmentary lineage in Pashtun tribal society

Prior historical and ethnographic accounts of Pashtun society provide important clues on the relative importance of segmentary lineage as a form of social organization. Here, we present some relevant discussion based on Beattie (2013).

Pervasive rivalry between patrilateral parallel male cousins (*taburwali*) offers an important reason why despite possessing some segmentary features, 'segmentary lineage theory is not in itself sufficient to explain political processes among the Mehsuds' (Beattie 2013: 11).

Beattie (2013: 22) goes on to note that:

From the point of view of social organization, one of the critical features shared by all the tribes, but exemplified by the Mehsuds in particular, included the weakness or virtual absence of political authority, and the fact that although they all possessed the framework of a segmentary lineage organization, they often did not organize themselves politically on segmentary lines. Instead, the prevalence of tarburwali tended to lead, it appears, to the emergence of small factional groups or 'alliance networks', membership of which crossed agnatic boundaries.

He further highlights how colonial administration found segmentary lineage as a poor predictor of tribal organization (Beattie 2013: 24):

the difficulty for the government officer was that both the segmentary lineage model and the factional model (and sometimes even the chiefly one) to some extent corresponded to reality [...] Sometimes tribal politics were shaped by clan and lineage membership, at other times they revolved around factional nuclei or even maliks; occasionally they expressed some kind of territorial identity.

# Appendix B: Appendix figures





Source: authors' construction based on data from New America (2021).



Figure B2: Visual lack of discontinuity in geographic, climatic, and historic factors

Source: authors' construction.



Figure B3: Conflict against the state and distance to FR border (smaller bins of size 2.5km each) (a) Number of incidents

(b) Number of deaths in incidents



(c) Number of injuries in incidents



Note: binned scatterplots (38 bins of size 2.5km each) of the unconditional relationship between conflict against the state and distance to the FR border. The y-axis reports the natural log of 1 plus the incidence of conflict against the state for each of our three measures. The x-axis reports the distance (in km) from the FR border for areas under FR and non-FR. The border itself is at km 0 with positive values indicating km inside the FR territory.

Source: authors' construction.


Figure B4: Conflict against the state (broader measure) and distance to FR border (a) Number of incidents

(b) Number of deaths in incidents



(c) Number of injuries in incidents



Note: binned scatterplots (38 bins of size 2.5km each) of the unconditional relationship between conflict against the state and distance to the FR border. The y-axis reports the natural log of 1 plus the incidence of conflict against the state for each of our three measures. The x-axis reports the distance (in km) from the FR border for areas under FR and non-FR. The border itself is at km 0 with positive values indicating km inside the FR territory.

Source: authors' construction.



Figure B5: SRD estimates for deaths and injuries after excluding spatial units close to the FR border (a) Number of deaths in incidents

(b) Number of injuries in incidents



Note: this figure shows estimated coefficients and confidence intervals for SRD specifications run on different restricted samples that successively exclude observations very close to the FR border. For example, 'donut\_1' excludes observations that are within 0.5km of the border, 'donut\_2' excludes observations within 0.4km of the border, etc.

Source: authors' construction.

Figure B6: Falsification test: moving the frontier rule border southwestward



Note: this figure shows the placebo FR border (black) that is based on moving the original border (red) 400km-by-550km in a southwest direction. Since Pakistan has a northeast-to-southwest orientation, this is equivalent to shifting the original FR border further inland into the country.

Source: authors' construction. The map is created through QGIS software.

Figure B7: Targeted attacks on tribal elders (Maliks)



Note: the map shows the attacks on tribal elders for the post-9/11 period (2006–18). Each black dot represents an attack against the tribal elders and the red line denotes the historical border (as of 1901) separating frontier agencies (left side of the border) from settled districts (right side of the border).

Source: authors' compilation. This map is created from GTD (2021) data through the QGIS application.

## Appendix C: Appendix tables

	Sample: observations within 50 km from FR border								
		Line	ar running variable in Eu	clidean distance to the l	oorder				
Dependent variable:	In(1+incidents	against state)	In(1+deaths in incid	dents against state)	In(1+injuries in inc	idents against state)			
	(1)	(2)	(3)	(4)	(5)	(6)			
Inside FR border	0.567***	0.563***	0.651***	0.609***	0.864***	0.850***			
	(0.088)	(0.086)	(0.101)	(0.099)	(0.101)	(0.100)			
Observations	1,118	1,105	1,118	1,105	1,118	1,105			
95% C.I.	[.427 ; .844]	[.427 ; .842]	[.512 ; .974]	[.473 ; .921]	[.72 ; 1.198]	[.722 ; 1.188]			
Controls	No	Yes	No	Yes	No	Yes			
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes			
BW-type	mserd	mserd	mserd	mserd	mserd	mserd			
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular			
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID			

Table B1: Frontier rule and conflict against the state using a broader definition of state targets

Note: this table shows results using a measure of violence against the state that is based on a **broader definition** of state targets beyond military installations and personnel. The **broader definition** also includes the following list of additional state affiliated targets: civil servants, teachers, doctors, judges, police, military personnel, parliamentarians, educational institutions, healthcare facilities, courts, roads, bridges, airports, electricity grids, and gas installations. The unit of observation is a 10km-by-10km grid cell. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Table B2: Frontier rule and conflict against the state using quadratic running variable

		:	Sample: observations wit	hin 50 km from FR bord	er	
		Quadi	ratic running variable in E	uclidean distance to the	border	
Dependent variable:	In(1+incidents	s against state)	In(1+deaths in inci	dents against state)	In(1+injuries in inci	dents against state)
	Sample: observations within 50 km from FR border   Quadratic running variable in Euclidean distance to the border   ariable: In(1+incidents against state) In(1+deaths in incidents against state) In(1+injuries in incidents against state)   (1) (2) (3) (4) (5)   (aer 0.229*** 0.245*** 0.321*** 0.315*** 0.672***   (0.083) (0.084) (0.122) (0.120) (0.157)   1,118 1,105 1,118 1,105 1,118   [.050; .420] [.067; .441] [.054; .598] [.058; .600] [.311; .830] [.40]   No Yes Yes Yes Yes Yes Yes Yes   Mo Yes Yes Yes Yes Yes Yes Yes   Mserd mserd mserd mserd mserd mserd Triangular	(6)				
Inside FR border	0.229***	0.245***	0.321***	0.315***	0.672***	0.743***
	(0.083)	(0.084)	(0.122)	(0.120)	(0.157)	(0.155)
Observations	1,118	1,105	1,118	1,105	1,118	1,105
95% C.I.	[.050 ; .420]	[.067 ; .441]	[.054 ; .598]	[.058 ; .600]	[.311 ; .830]	[.331 ; .862]
Controls	No	Yes	No	Yes	No	Yes
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes
BW-type	mserd	mserd	mserd	mserd	mserd	mserd
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID

Note: the unit of observation is a 10km-by-10km grid cell. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a quadratic polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

		:	Sample: observations withi	in 50 km from FR borde	er	
Dependent variable:	In(1+incidents	against state)	In(1+deaths in in stat	icidents against te)	In(1+injuries in ir stat	ncidents against re)
Dependent variable: Inside FR border Observations R <sup>2</sup>	(1)	(2)	(3)	(4)	(5)	(6)
		F	Panel A: Linear polynomial	l in latitude and longitud	de	
Inside FR border	0.128	0.091	0.202	0.131	0.187	0.118
	(0.039)***	(0.044)**	(0.059)***	(0.067)*	(0.061)***	(0.067)*
	[0.060]**	[0.058]	[0.088]**	[0.081]	[0.087]**	[0.076]
Observations	1,118	1,105	1,118	1,105	1,118	1,105
R <sup>2</sup>	0.188	0.221	0.162	0.189	0.155	0.181
		Pa	nel B: Quadratic polynomi	al in latitude and longit	ude	
Inside FR border	0.130	0.126	0.200	0.180	0.182	0.158
	(0.042)***	(0.047)***	(0.063)***	(0.070)**	(0.064)***	(0.070)**
	[0.063]**	[0.058]**	[0.089]**	[0.081]**	[0.087]**	[0.078]**
Observations	1,118	1,105	1,118	1,105	1,118	1,105
R <sup>2</sup>	0.198	0.237	0.174	0.203	0.166	0.191
Controls	No	Yes	No	Yes	No	Yes
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes

Table B3: Frontier rule and conflict against the state using alternative functional forms for the running variable

Note: the unit of observation is a 10km-by-10km grid cell. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state; all parameterized as ln(1 + x). Panel A includes a linear polynomial in latitude and longitude as the running variable. Panel B includes a quadratic polynomial in latitude and longitude as the running variable. All regressions include border segment fixed effects that divide the FR border into four equal size segments. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Robust standard errors are in parentheses, and standard errors corrected for spatial dependence are in brackets. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

			Sample: observations wit	hin 50 km from FR borde	r	
		Lin	ear running variable in Eu	clidean distance to the bo	order	
Dependent variable:	In(1+incidents	s against state)	In(1+deaths in inci	dents against state)	In(1+injuries in inci	dents against state)
	(1)	(2)	(3)	(4)	(5)	(6)
			Panel A: 15 km bandwid	Ith used for RD estimate		
Inside FR border	0.136**	0.139**	0.194**	0.186**	0.277***	0.272***
	(0.056)	(0.055)	(0.087)	(0.084)	(0.074)	(0.072)
Observations	1,118	1,105	1,118	1,105	1,118	1,105
95% C.I.	[.04 ; .366]	[.049 ; .373]	[.039 ; .519]	[.037 ; .505]	[.203 ; .626]	[.202 ; .618]
			Panel B: 12 km bandwid	oth used for RD'estimate		
Inside FR border	0.162***	0.170***	0.230***	0.209**	0.340***	0.329***
	(0.055)	(0.055)	(0.083)	(0.081)	(0.072)	(0.070)
Observations	1,118	1,105	1,118	1,105	1,118	1,105
95% C.I.	[.063 ; .403]	[.073 ; .412]	[.072 ; .566]	[.063 ; .552]	[.265 ; .722]	[.259 ; .716]
			Panel C: 10 km bandwid	dth used for RD estimate		
Inside FR border	0.250***	0.260***	0.372***	0.346***	0.501***	0.486***
	(0.053)	(0.052)	(0.075)	(0.076)	(0.067)	(0.067)
Observations	1,118	1,105	1,118	1,105	1,118	1,105
95% C.I.	[.156 ; .503]	[.165 ; .511]	[.168 ; .702]	[.160 ; .695]	[.410 ; .903]	[.397 ; .902]
Controls	No	Yes	No	Yes	No	Yes
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes
BW-type	manual	manual	manual	manual	manual	manual
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID

Table B4: Frontier rule and conflict against the state using alternative manually chosen bandwidths

Note: the unit of observation is a 10km-by-10km grid cell. Panel A manually imposes a bandwidth of 15 km either side of the FR border for the RD estimate. Panel B uses a bandwidth of 12 km either side of the FR border for the RD estimate. Finally, Panel C imposes a bandwidth of 10 km either side of the FR border for the RD estimate. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and border segment fixed effects. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Table B5: Frontier rule and conflict against the state using alternative kernel weights

			Sample: observations wit	within 50 km from FR borderEuclidean distance to the borderincidents against state)In(1+injuries in incidents against state)(4)(5)(6)0.181**0.568***0.526***(0.087)(0.075)(0.075)1,1051,1181,105[.035 ; .426][.469 ; .805][.425 ; .759]YesYesYesYesYesYesgenechnikovEpanechnikovEpanechnikovEpanechnikovSegment IDSegment ID		
		Line	ar running variable in Eu	clidean distance to the b	order	
Dependent variable:	In(1+incidents	s against state)	In(1+deaths in inci	dents against state)	In(1+injuries in inci	dents against state)
	(1)	(2)	(3)	(4)	(5)	(6)
Inside FR border	0.254***	0.267***	0.275***	0.181**	0.568***	0.526***
	(0.056)	(0.057)	(0.082)	(0.087)	(0.075)	(0.075)
Observations	1,118	1,105	1,118	1,105	1,118	1,105
95% C.I.	[.178 ; .422]	[.193 ; .444]	[.146 ; .513]	[.035 ; .426]	[.469 ; .805]	[.425 ; .759]
Controls	No	Yes	No	Yes	No	Yes
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes
BW-type	mserd	mserd	mserd	mserd	mserd	mserd
Kernel	Epanechnikov	Epanechnikov	Epanechnikov	Epanechnikov	Epanechnikov	Epanechnikov
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID

Note: the unit of observation is a 10km-by-10km grid cell. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. They also use Epanechikov kernel weights (as opposed to Triangular kernel weights) for weighting observations closer to the running variable cutoff. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

			Sample: observations w Linear running variable in	ithin 50 km from the <b>placebo</b> Euclidean distance to the <b>pla</b>	FR border I <b>cebo</b> border	
Dependent variable:	In(1+incidents	against state)	In(1+deaths in inci	dents against state)	In(1+injuries in inc	idents against state)
	(1)	(2)	(3)	(4)	(5)	(6)
Inside <b>placebo</b> FR border	-0.789	-0.839	-1.303	-1.257	-0.049	0.308
	(0.760)	(0.806)	(1.261)	(1.197)	(0.106)	(0.775)
Observations	1,216	1,216	1,216	1,216	1,216	1,216
95% C.I.	[-2.439 ; .675]	[-2.567 ; .717]	[-4.181 ; 1.290]	[-4.025 ; 1.194]	[259 ; .191]	[-1.187 ; 1.832]
Controls	No	Yes	No	Yes	No	Yes
Segment FE	Yes	Yes	Yes	Yes	Yes	Yes
BW-type	mserd	mserd	mserd	mserd	mserd	mserd
Kernel	Triangular	Triangular	Triangular	Triangular	Triangular	Triangular
Clustering	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID	Segment_ID

Table B6: Falsification test based on moving the frontier rule border in a southwest direction

Note: this table shows the results for a **placebo** FR border that is based on moving the original border 400km-by-550km in a Southwest direction. Since Pakistan has a Northeast to Southwest orientation, this is equivalent to shifting the original FR border further inland into the country. The unit of observation is a 10km-by-10km grid cell. In columns 1–2, the outcome variable is the number of conflict incidents against the state; in columns 3–4, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5–6, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Since there is ambiguity in the treatment status of grid cells very close to the **placebo** border, all regressions also exclude grid cells that are within a distance of 0.5km from it. Columns 2, 4, and 6 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

					Sample: observ	ations with	in 60 km fro	m FR border				
				Lin	ear running varia	able in Euc	lidean distar	nce to the bo	rder			
			Pre-	9/11					Post-9	9/11		
Depende nt variable:	ln(1+iı again	ncidents st state)	In(1+de incidents sta	eaths in s against ite)	In(1+in incident sta	juries in s against ite)	In(1+ir agains	cidents st state)	In(1+de incident sta	eaths in s against ate)	In(1+inj incidents sta	uries in s against ate)
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Inside FR border	-0.008	-0.010	-0.011	-0.012	-0.007	-0.008	0.281***	0.297***	0.302***	0.191**	0.611***	0.564
	(0.006)	(0.006)	(0.008)	(0.008)	(0.005)	(0.005)	(0.053)	(0.053)	(0.077)	(0.084)	(0.071)	(0.071 )
Obs.	1,288	1,271	1,288	1,271	1,288	1,271	1,288	1,271	1,288	1,271	1,288	1,271
95% C.I.	[021 ; .006]	[023 ; .005]	[027 ; .007]	[030 ; .006]	[018 ; .005]	[019 ; .004]	[.208 ; .438]	[.229 ; .465]	[.182 ; .517]	[.034 ; .410]	[.520 ; .832]	[.472 ; .788]
Controls Segment	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes	No	Yes
FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
BW-type	mserd	mserd Triangula	mserd	mserd Triangula	mserd	mserd	mserd	mserd Triangula	mserd	mserd Triangula	mserd Triangula	mserd Triang
Kernel	Triangular	· r	Triangular	r	Triangular	Triangular	Triangular	r	Triangular	r	r	ular
Cluste	Segment	Segment	Segment S	egment	Segment S	egment	Segment S	egment	Segment S	Segment	Segment	Segment
ring	_ID	_ID	_ID	_ID	_ID	_ID	_ID	_ID	_ID	_ID	_ID	_ID

Table B7: Frontier rule and conflict against the state in the pre- and post-9/11 era

Note: the unit of observation is a 10km-by-10km grid cell. Columns 1–6 restrict the sample to the period prior to 9/11 from 1970 to 2000 and columns 7–12 restrict the sample to the period after 9/11 from 2001 to 2018. In columns 1–2 and 7–8, the outcome variable is the number of conflict incidents against the state; in columns 3–4 and 9–10, the dependent variable is the number of deaths in conflict incidents against the state; and in columns 5-6 and 11-12, the dependent variable is the number of injuries in conflict incidents against the state, all parameterized as ln(1 + x). All regressions include a linear polynomial in distance to the border and 20 km border segment fixed effects. Columns 2, 4, 6, 8, 10, and 12 also include the following set of controls: ruggedness, topography, slope, precipitation, temperature, wheat suitability, pre-FR major conflict incidence, and pre-FR population density. Standard errors, clustered at the border segment ID level, are reported in parentheses. \*, \*\*, and \*\*\* indicate significance at the 10%, 5%, and 1% levels.

Table B8: Conflict incidents against the state by specific origin

	197	0–2018	197	0–2000	2001–18	
Militant outfit	Count	Per cent	Count	Per cent	Count	Per cent
	Loca	I				
Tehrik-i-Taliban Pakistan (TTP)	875	12.04	0	0.00	875	12.87
Baluchistan-based militants	610	8.40	49	10.45	561	8.25
Political militants wings	175	2.41	34	7.25	141	2.07
Lashkar-e-Jhangvi	294	4.05	1	0.21	293	4.31
Different local jihadi organization	1378	18.97	104	22.17	1274	18.74
Other militant organizations	1874	25.79	101	21.54	1773	26.09
	Foreig	jn				
Haqqani network	2	0.03	0	0.00	2	0.03
Al-Qaida	31	0.43	0	0.00	31	0.46
	Unknow	wn	-		-	
Unknown	2027	27.90	180	38.38	1847	27.17
Total	7266	100	469	100	6797	100

Note: the **Different local jihadi organizations** category includes groups like the Sipah-e-Sahaba, Hizb-I-Islami, Tehrik-e-Nafaz-e-Shariat-e-Mohammadi, Lashkar-e-Islam, Ansarul Islam, Jaish-e-Islam, Ja

Source: based on authors' calculations.